# Information and Optimal Trading Strategies with Dark Pools<sup>\*</sup>

Anna Bayona<sup>1</sup>

Ariadna Dumitrescu<sup>2</sup>

Carolina Manzano<sup>3</sup>

October 2017

#### Abstract

We study how asymmetric information affects market participants' choice of trading venue (either an exchange or dark pool), and the optimal submission strategies in a sequential trading game. The exchange is organized as a fully transparent limit order book, and the dark pool is an opaque venue where orders are continuously executed at the midpoint of the bid and ask prices that prevail in the exchange. We find that, when the limit order book conveys no information, rational uninformed traders never trade in the dark pool due to price risk. However, price risk may be reduced when the information in the book induces an uninformed buyer (seller) to believe that the value of the asset is high (low) since the order was previously submitted by an informed buyer (seller). Adding a dark pool alongside an exchange may divert the informed trader from the exchange to the dark pool if the execution risk in the dark is sufficiently low. An uninformed trader only goes to the dark if the limit order book is sufficiently informative and price risk is low. We show that adding a dark pool alongside an exchange reduces price informativeness and increases the expected welfare of rational traders. Its effects on market liquidity and trading volume depend on stock market characteristics since these determine whether traders supply, demand or do not provide liquidity in the exchange when the dark pool is unavailable.

*Keywords:* trading venues, dark liquidity, limit order book, price risk, adverse selection *JEL codes:* G12, G14, G18

<sup>&</sup>lt;sup>\*</sup>The authors would like to thank Julio Crego, Laurence Lescourret, Roberto Pascual, and seminar participants at Finance Theory Group Summer School 2017, ERMAS 2017, CEPR European Summer Symposium in Financial Markets 2017, Finance Forum 2017, CEPR-Imperial-Plato Conference, ESSEC Business School, Universitat de les Illes Baleares and ESADE Business School. Any errors are our own responsibility.

<sup>&</sup>lt;sup>1</sup>Anna Bayona, ESADE Business School, Av. Pedralbes, 60-62, Barcelona, 08034, Spain. *Tel:* (34) 932 806 162, *e-mail:* anna.bayona@esade.edu

<sup>&</sup>lt;sup>2</sup>Ariadna Dumitrescu, ESADE Business School, Av. Pedralbes, 60-62, Barcelona, 08034, Spain. *Tel:* (34) 934 952 191, *e-mail:* ariadna.dumitrescu@esade.edu

<sup>&</sup>lt;sup>3</sup>Carolina Manzano: Universitat Rovira i Virgili and CREIP. Av. de la Universitat 1, Reus, 43204, Spain. *Tel:* (34) 977 758 914, *e-mail:* carolina.manzano@urv.cat

## 1 Introduction

In today's financial markets, traders have access to different types of trading venues, which differ in their level of transparency. In addition to the traditional exchanges (lit markets), traders also have access to dark pools - "trading venues or mechanisms containing anonymous, non-displayed liquidity that is available for execution" (Banks, 2014). Dark pools grew as a result of technological innovations and Reg NMS regulation in 2005.<sup>1</sup> Their current consolidated trading volume in US equity markets is around 14%, while in European equity markets is around 9.1% (Rosenblatt Securities Inc.). Our paper studies how the existence of a dark pool affects traders' optimal submission strategies in a sequential trading game with asymmetric information. Informed and uninformed traders face a simultaneous choice of trading venue and order type in the exchange (market order or limit order) or refrain from trading. We model the competition between an exchange (that is organized as a limit order book) and a dark pool in the presence of asymmetric information. Our model therefore allows us to understand the role played by information in the price discovery process and the strategic behavior of traders in the presence of asymmetric information but also, unlike other papers in the literature, we study the optimal venue and type of order decision, and how the leakage of information affects this decision.

Our main finding is that adding a dark pool alongside an exchange decreases price informativeness in the exchange in the first period and improves expected profits of both informed and uninformed traders. The effect on expected inside spread and trading volume depends on stock market characteristics such as liquidity, volatility, adverse selection, tick size etc. In addition, adding a dark pool alongside an exchange may switch the optimal strategies of each type of traders. Thus, in the first period, the informed trader may divert from the exchange to the dark pool if the execution risk in the dark pool is sufficiently low. The uninformed trader does not go to the dark pool when the limit order book contains no information since price risk is too high. However, adding a dark pool alongside an exchange may switch the optimal strategy of the uninformed trader from no trade to placing a limit order in the exchange due to the reduction in adverse selection. An uninformed trader only goes to the dark when the limit order book is sufficiently informative and when price risk is low. The stock market characteristics determine also if there is or not segmentation of the informed and uninformed order flow in the two venues. These results are very important given the current policy debate on the impact of the dark pools on price discovery, market liquidity and order flow segmentation.

Our model reflects the main characteristics of today's financial markets. The exchange is organized as a fully transparent limit order book with a discrete price grid. Despite the fact that there exist many types of dark pools, our modeling captures two of their main features: (1) no pre-trade transparency. Dark pools are completely opaque in the sense that do not quote the liquidity that is available and this makes execution uncertain; (2) they do not determine prices and derive their

<sup>&</sup>lt;sup>1</sup>The policy debate on the regulation of dark pools is currently very active (see, for example, U.S. Securities and Exchange Commission, 2015).

price from those prevailing in the exchange as the midpoint between the best bid and ask prices at any point in time (if the order is executed). This type of pricing is typical of dark pools which are owned by agency brokers or exchanges and represent 57% of the consolidated dark trading volume (Buti et al. 2017). Traders may be of two different types: rational or liquidity traders. Rational traders strategically choose whether or not to trade, and if they trade they simultaneously choose the venue and the type of order (in the exchange both market and limit orders are available) given their information at each point in time. Rational traders are (privately) informed if they know the liquidation value of the asset, and (privately) uninformed traders if they only know the distribution of the liquidation value of the asset conditional on the information provided by the book. Since the limit order book is fully transparent, information in the book is public and available to all types of traders.

The timing of the model is as follows. First, the liquidation value of the asset is realized. Second, there are two periods of trading. In each period, a new trader may arrive to the market. Rational traders observe the state of the limit order book and strategically choose an order submission strategy that maximizes expected profits given the information set at each period. Liquidity traders always trade based on their exogenous liquidity needs, and only submit market orders to the exchange to ensure execution. Third, if in the first period a trader had submitted a dark pool order and this order was not executed in the dark, then the order returns to the exchange. Fourth, the liquidation value of the asset is made public and the trading game is over. Since our model can be represented by a sequential game of incomplete information, the equilibrium concept used is the Perfect Bayesian Equilibrium. To the best of our knowledge, we are the first to model the competition between an exchange that is organized as a limit order book and a dark pool in the presence of asymmetric information.

To understand the effects of adding a dark pool alongside the exchange, we first discuss the equilibrium in the benchmark model where traders do not have access to the dark pool.<sup>2</sup> We solve the model backwards. In the last trading period, limit orders are not chosen since they will not be executed. An informed trader always chooses a market order since it gives positive profits. An uninformed trader obtains information about the state of the book and updates his beliefs about the value of the asset. An uninformed trader selects a market order if he strongly believes that an informed trader in the previous period had chosen an order of the same direction, which indicates that the value of the asset is favorable. Otherwise, an uninformed trader refrains from trading. In the first trading period, an informed trader chooses a market order over a limit order when he prefers immediacy to the potential price improvement provided by a limit order. An uninformed trader selects a limit order instead of not trading when there is a high probability that his order will be executed against the order of a liquidity trader instead of an informed trader (low adverse selection). If his limit order is executed against a market order of the opposite sign submitted by an informed trader, then it reveals that the value of the asset is disadvantageous (high adverse selection). Even

 $<sup>^{2}</sup>$ In the benchmark model the third stage described in the previous paragraph is dropped.

though the optimal strategies profiles are unique in the first trading period, multiple equilibria may exist in the case that in the first trading period informed traders choose market orders and uninformed traders choose not to trade.

Adding a dark pool not only enlarges traders' strategies set but also may induce a substitution of trading venue, order type, and market participation in relation to when the dark pool is unavailable. In the second round of trading, the optimal trading strategy depends on the state of the LOB and the assessment of the expected profits. An informed trader chooses between a market order or a dark order depending on the trade-off between price improvement and execution risk in the dark versus immediate execution. In the second round of trading, uninformed traders refrain from trading if the LOB is sufficiently uninformative. However, if the LOB indicates the value of the asset is favorable, uninformed traders choose between market orders and dark orders. Whether there is segmentation of the order flow in the second round of trading (equilibria where informed and uninformed participate both in the lit and dark or equilibria where only one type of traders participates in the dark) depends on stock characteristics.

In the first trading period, even if execution risk in the dark pool is high, informed traders tend to replace market orders by dark orders when they can take advantage of the price improvement. The expected profit from submitting a dark order is the largest when the discount factor is high and when there is a small probability that the market moves against the trader. As the execution risk in the dark reduces, strategy profiles in which informed traders submit market orders cannot be anymore an equilibrium of the game, and informed traders decide to go to the dark. Despite of the fact that uninformed traders do not select to go to the dark pool when the limit order book is uninformative (due to adverse selection and price risk in the first trading period), adding a dark pool alongside an exchange affects the optimal submission strategies of the uninformed trader even in the first period. Uninformed traders may switch from no trade to trade in the exchange using limit orders. This is because the mere existence of the dark market offers the possibility to informed traders in the second period to migrate from the exchange to the dark, and consequently reduce the adverse selection which induces uninformed traders to participate in the market submitting limit orders. Despite the fact that optimal strategies profiles are unique in the first trading period, multiple equilibria may exist unless in the first trading period informed traders choose dark pool orders and uninformed traders choose limit orders.

In terms of market quality and welfare, we find that the addition of a dark pool alongside an exchange decreases price informativeness but it increases the expected profits of rational traders (both informed and uninformed). The effect of adding a dark pool on the expected inside spread and trading volume depends on stock market characteristics. These characteristics determine if a trader who migrates to the dark pool was a supplier or a consumer of liquidity in the exchange (when the dark pool was unavailable) and therefore whether the migration to the dark pool has a positive or negative effect on market liquidity and trading volume. These theoretical results help reconcile the positive and negative effects the existence of dark pools has on the market performance

of competing exchanges found in the empirical studies, and provides new empirical predictions for cross-sectional analysis.

Our paper is closely related to the theoretical analysis which analyzes the effects of adding a dark pool alongside an exchange on market performance. This literature shows that the impact of the dark pool on price discovery is ambiguous. On the one hand, there is a strand of literature that studies the competition between a dealer market and a dark pool (or crossing network) in the presence of asymmetric information. Zhu (2014), using a Glosten and Milgrom (1985) type model, finds that adding a dark pool alongside an exchange concentrates price-relevant information into the exchange and improves price discovery. This is due to the fact that a continuum of informed traders receive the same perfect signal and trade simultaneously on the same signal.<sup>3</sup> Therefore, when they submit orders to the dark pool, their execution probability reduces as all submit orders on the same side. Thus, Zhu (2014) shows that the informed traders submit orders to the dark pool only if the uncertainty of the asset value is very high. Ye (2011), using a Kyle (1985) type model, finds that a dark pool reduces price discovery and volatility. Note that these two models impose a different market structure: Ye (2011) only allows the informed trader to select trading venue, while Zhu (2014) allows both informed and liquidity traders to select the venue to trade.

Competition between dealer markets and other forms of exchange, such as passive crossing networks (similar to dark pools), and in the presence of asymmetric information has also been analyzed by Hendershott and Mendelson (2000) and Degryse et al. (2009). Hendershott and Mendelson (2000) in a static setup find that a crossing network imposes positive liquidity externalities and negative crowding externalities on each other and therefore have ambiguous effects on spread. Degryse et al. (2009) show that the same positive and negative externalities are preserved in a dynamic setup and analyze how welfare and the order flow dynamics depend on the degree of market transparency.

On the other hand, Buti et al. (2017) model competition between a fully transparent limit order book and a dark pool without asymmetric information regarding the asset value. They show that the welfare effects of adding a dark pool are negative if the initial book is illiquid, while when book liquidity increases, large traders are better off and small traders are worst off. They also find that the market share of the dark pool is higher when the depth of the limit order book is high, when the spread of the limit order book is narrow, when the tick size is large, and when traders seek protection from price impact.

We build a limit order book model with asymmetric information that complements the models of Zhu (2014) and Buti et al. (2017). We model a limit order book model where traders have asymmetric information and common valuation, while Buti et al. (2017) consider a model of symmetric information and private valuations. The existence of private information in our model permits us to analyse the informational role of prices, the adverse selection problem and their impact on the segmentation of the order flow, as well as to study the traders'dynamic strategies as in Buti et al. (2017). Compared with Zhu (2014) who models competition between a dealer market (where

 $<sup>^{3}</sup>$ Note that Zhu (2014) models a market with a large number of informed strategic traders. As Kyle (1989) points out this does not reflect the situation of real financial markets.

traders submit market orders) and a dark pool with asymmetric information, we model a limit order book where traders can submit both market and limit orders to the exchange. Moreover, trading takes place sequentially in our model and this together with the different market structure leads to exactly opposite results in terms of price discovery. Note that in our model the segmentation of the order flow arises endogenously and it depends on the stock market characteristics. We show that when the order book is not informative the informed traders might migrate to the dark pool while uninformed stay in the lit market and therefore the informed traders do not contribute to the price discovery in the lit market.

Our research is also related to two broader strands of literature that the previously mentioned papers tackle: competition between integrated and segmented markets, and traders' optimal order submission strategies. On the one hand, early theoretical papers involving multiple trading venues (see, for instance, Pagano, 1989; and Chowdry and Nanda, 1991) show that there is a natural tendency towards agglomeration since liquidity increases due to scale, and it is beneficial. This tendency may be offset by the presence of frictions, trading costs, informational barriers or regulatory obstacles. Since market fragmentation is associated with the surge of venues with different degrees of transparency, a part of this literature relates to fragmentation and market transparency.<sup>4</sup> Concerning the visibility of market quotes, for instance, Biais (1993) and Frutos and Manzano (2002) compare centralized and fragmented markets and show that the ability to observe price setters' quotes affects spreads and market participants' welfare. In relation to the disclosure of post-trade information, Madhavan (1995) shows that delaying disclosure benefits large traders who place multiple trades. Frutos and Manzano (2005) in a two-stage trade model show that opaqueness increases competition among dealers to attract valuable order flow, leading to better prices for investors in the first period, while the effect on market participants' welfare in second period is ambiguous.

On the other hand, there is also a large literature that studies limit order markets and traders' optimal order submission strategies with models of asymmetric information. The early static models of limit order markets assume that informed traders only use market orders, while uninformed traders or liquidity traders only use limit orders. Angel (1994), Easley and O'Hara (1991) and Harris (1998) model an informed investor's order placement strategy in choosing between market and limit orders. They argue that informed traders are more likely to use market orders and rarely use limit orders. Also, Glosten (1994), Rock (1996), Seppi (1997), and Biais et al. (2000) argue that informed traders to profit from their private information. In contrast, Chakravarty and Holden (1995) and Kaniel and Liu (2006), who analyze informed traders' choice between limit orders and market orders, find that informed traders may prefer limit orders since they may actually convey less information than market orders. Finally, Parlour (1998), Foucault (1999), Foucault, Kadan and Kandel (2005), Goettler et al. (2009), Rosu (2009), Van Achter (2008), and Rosu (2012) introduce dynamic models that allow informed and uninformed traders to determine their optimal choice of order type in limit order markets. Empirical studies find that both informed

 $<sup>^{4}</sup>$ See also Gomber et al. (2016) for a review of the consolidation versus fragmentation of markets.

and uninformed traders use a mixture of market orders and limit orders (see for example Biais et al. 1995; Kavajecz and Odders-White 2004; Anand et al., 2005). Bloomfield et al. (2005) conduct a laboratory experiment and find that informed traders use both market and limit orders. They use market orders earlier in the trading period to profit from their private information, and then (as the prices get closer to true value) they use their private information to switch to limit orders to earn the bid-ask spread. Uninformed traders use limit orders early but then switch to market orders to meet their liquidity targets (they are liquidity traders). They also find that informed traders' submission patterns are more sensitive to changes in market transparency. We build on this literature by developing a model in which rational traders can decide simultaneously the venue in which they want to trade and their optimal order submission strategy.

The paper is organized as follows. Section 2 presents the model. Section 3 analyzes the equilibrium in benchmark model without dark pool. Section 4 presents the equilibrium in the full model where rational traders have access to the dark pool. Section 5 analyzes how welfare and market quality change when we add a dark pool alongside an exchange and section 6 provides the empirical implications of these results. Section 7 concludes. Proofs are presented in the Appendices.

## 2 Model

We consider an economy in which a single risky asset is traded. The liquidation value of the asset,  $\tilde{v}$ , may take two values,  $v \in \{v^H, v^L\}$ , with equal probabilities,  $\mu$  is the unconditional mean of  $\tilde{v}$ , and  $\sigma > 0$  its standard deviation. The asset may be traded in two venues: an exchange or a dark pool.

The exchange is organized as a limit order book (hereafter LOB). We assume that the initial LOB has two prices on each side of the book:  $A_1^1$ ,  $A_1^2$ ,  $B_1^1$ ,  $B_1^2$ , such that  $v^L < B_1^2 < B_1^1 < A_1^1 < A_1^2 < v^H$ . We assume that prices are placed on a grid and that the following relationships hold:

$$A_1^1 = \mu + k_1 \tau, \ A_1^2 = \mu + k_2 \tau, \ v^H = \mu + k_3 \tau,$$
  
$$B_1^1 = \mu - k_1 \tau, \ B_1^2 = \mu - k_2 \tau, \ v^L = \mu - k_3 \tau,$$

with  $1 \leq k_1 < k_2 < k_3$ , where  $k_1$  and  $k_2$  are natural numbers, and  $\tau$  is the tick size (i.e., the minimum price increment that traders are allowed to quote over the existing price). Note that  $k_3\tau = \sigma$ , and therefore,  $k_3$  is a real number. For simplicity we assume that the depth of the LOB at each bid and ask price is equal to 1, and that the LOB follows price and time priority rules.<sup>5</sup> The LOB is fully transparent (i.e., all the information in the LOB is available to all market participants at any point in time). There are no transaction costs.

The dark pool is completely opaque in the sense that an order submitted to the dark pool is not observable to anyone but the trader who submitted it. We assume that the dark pool has

 $<sup>{}^{5}</sup>$ First, the order with the best price is executed. Second, among the orders with the same price, they are executed in order of arrival.

t = 0	t = 1	t = 2	t = 3	t = 4
Nature draws	• A trader may arrive	• A new trader may arrive	If at $t = 1$ a DO	Liquidation value
$v \in \left\{ v^H, v^L \right\}$	• Rational traders choose their optimal submission strategies based on each trader's information set	• Rational traders choose their optimal submission strategies based on each trader's information set	was submitted and not attended the order returns to the lit market	is made public and the trading game is over
	• Liquidity traders trade according to their liquidity needs	• Liquidity traders trade according to their liquidity needs		

Figure 1: Time line of the trading game when traders have access to the dark pool

an execution probability  $\theta \in [0,1]$  that is exogenous and does not change in time. If an order is submitted to the dark pool and it is attended at time t then the execution price is equal to the midpoint of the exchange at time t:  $\frac{A_t^1 + B_t^1}{2}$ . If the order is not attended in the dark pool then it returns to the exchange at t + 2.<sup>6</sup>

The sequence of events described also in Figure (1) is as follows:

**Date t=0:** The liquidation value of the asset  $\tilde{v}$  is realized.

Dates t=1, 2: In each date, a *new* trader may arrive to the market and may either trade 0 or 1 unit of the asset. An informed trader observes the liquidation value of the asset. The state of the LOB at the beginning of each date is public information. A rational trader chooses an order submission strategy that maximizes his expected profits given the information set at each date. Liquidity traders always trade based on their exogenous liquidity needs. All traders may trade one unit of the asset.

**Date t=3:** If at t = 1 a rational trader submitted a dark pool order and this order was not attended, then the order returns to the exchange as a market order.

Date t=4: The liquidation value of the asset is made public and the trading game is over.

Figure 2 illustrates the tree of events related to the first trading period.<sup>7</sup> There are two possible types of traders: rational traders or liquidity traders. All traders are risk neutral. A rational trader arrives to the market with probability  $\lambda > 0$ , a liquidity trader arrives with probability  $\eta > 0$ , and no trader arrives with probability  $1 - \lambda - \eta \ge 0$ . Rational traders may be either (privately) informed if they have perfect information about the liquidation value of the asset (with probability  $\pi$ ), or (privately) uninformed if they only know the distribution of the liquidation value of the asset (with probability  $1 - \pi$ ). An informed trader buys whenever he observes  $v = v^H$  (henceforth IH), and sells whenever he observes  $v = v^L$  (henceforth IL). An uninformed trader is a buyer with

<sup>&</sup>lt;sup>6</sup>Our assumptions are motivated by agency-broker or exchanged-owned dark pools. In these dark pools, price discovery does not take place since, if an order is executed, the price is equal to the midpoint of the National Best Bid and Offer (in this model the lit exchange).

 $<sup>^7\</sup>mathrm{A}$  similar tree of events for the second trading period could be drawn.



Figure 2: Tree of events of the first trading period.

probability  $\frac{1}{2}$  (henceforth UB) and a seller with probability  $\frac{1}{2}$  (henceforth US). There is also a discount factor,  $\delta \in [0, 1]$ , that is common across traders and periods.

Liquidity traders buy (with probability  $\frac{1}{2}$ ) or sell (with probability  $\frac{1}{2}$ ) for liquidity or hedging needs. Hence, in order to ensure immediate execution they trade in the exchange and always submit market orders. Rational traders trade up to one unit of the risky asset and choose an order submission strategy that maximizes their expected profits conditional on their information set at each date,  $I_t$ , which includes information about the liquidation value of the asset and about the state of the LOB. Rational traders simultaneously select whether not to trade (NT) or to trade, the trading venue (exchange or dark pool, DO), and the order type in the exchange (market order, MO, or limit order, LO). Consequently, the possible strategies of a rational trader (both informed and uninformed) are

$$\mathbb{O}_D = \{MO, LO, DO, NT\},\tag{1}$$

where B in front of an order type denotes a buy order and S denotes a sell order. Note that the

direction of trade for informed traders is endogenous since it depends on their private information.

For each possible order type, we next examine its characteristics and the associated profits for a rational trader that submits it. Internet Appendix I describes the expected profits of all traders, at all times, for all the possible states of the LOB.<sup>8</sup> Denote the profits of a particular order as  $\Pi^R_{\mathcal{O},t}$ , where superscript R denotes that the order comes from a rational trader (either informed, I, or uninformed, U); subscript  $\mathcal{O}$  is the order type  $\mathcal{O} \in \mathbb{O}_D$  defined in (1); and subscript t is the date when the order is submitted. If profits are strictly positive then a rational trader can choose between a MO or LO, and if available, or a DO. Otherwise, he chooses not to trade (NT).

• Market order (MO): Market orders are executed immediately at the given best available ask/bid prices. The expected profits of a buy market order at date t are

$$\mathbb{E}\left(\Pi_{BMO,t}^{R}|I_{t}\right) = \mathbb{E}\left(\widetilde{v}|I_{t}\right) - A_{t}^{1},$$

and the expected profits of a sell market order are

$$\mathbb{E}\left(\Pi_{SMO,t}^{R}|I_{t}\right) = B_{t}^{1} - \mathbb{E}\left(\widetilde{v}|I_{t}\right).$$

• Limit orders (LO): A limit order that improves the current market price may be executed in the next period if a market order of the opposite sign hits the limit order. Therefore, the expected profits from a limit order are discounted by  $\delta$ . Thus, limit orders provide better prices than market orders but exhibit execution risk. We assume that a LO always improves the price by one tick because: (i) it is never optimal for the trader to improve the price by more than one tick since it reduces his profits. (ii) it is never optimal for the trader to submit a non-improving LO since the order is not executed (the order goes to the queue, and due to the time priority, it is not executed), and the trader obtains zero profits. Hence, the expected profits of a buy limit order at date t are

$$\mathbb{E}\left(\Pi_{BLO,t}^{R}|I_{t}\right) = p_{BLO,t}^{R}\left(I_{t}\right)\delta(\mathbb{E}\left(\widetilde{v}|I_{t}\right) - B_{t}^{1} - \tau),$$

and of a sell limit order at date t is

$$\mathbb{E}\left(\Pi_{SLO,t}^{R}|I_{t}\right) = p_{SLO,t}^{R}\left(I_{t}\right)\delta(A_{t}^{1} - \tau - \mathbb{E}\left(\widetilde{v}|I_{t}\right)),$$

where  $p_{BLO,t}^{R}$   $(p_{SLO,t}^{R})$  is the probability of execution of a buy (sell) *LO* submitted by a rational trader at time *t*, respectively.

• Dark orders (DO): With probability  $\theta$  an order submitted to the dark pool is attended (executed), and with probability  $(1 - \theta)$  it is not attended. Since no new trader arrives in the

<sup>&</sup>lt;sup>8</sup>The Internet Appendix I is available from the authors upon request.

market at t = 3, 4, an order that returns to the exchange from dark pool at t+2 will be either a *MO* or *NT* since the probability of execution of a *LO* at t = 3 is equal to 0. Therefore, the expected profits of a buy dark order submitted at time t are:

$$\mathbb{E}\left(\Pi_{BDO,t}^{R}|I_{t}\right) = \theta\left(\mathbb{E}\left(\widetilde{v}|I_{t}\right) - \frac{A_{t}^{1} + B_{t}^{1}}{2}\right) + \max\left\{(1-\theta)\delta^{2}\mathbb{E}\left(\Pi_{BMO,t+2}^{R}|I_{t}\right), 0\right\}$$

and for a sell dark order submitted at time t, the expected profits equal

$$\mathbb{E}\left(\Pi_{SDO,t}^{R}|I_{t}\right) = \theta\left(\frac{A_{t}^{1} + B_{t}^{1}}{2} - \mathbb{E}\left(\widetilde{v}|I_{t}\right)\right) + \max\left\{(1-\theta)\delta^{2}\mathbb{E}\left(\Pi_{SMO,t+2}^{R}|I_{t}\right), 0\right\}.$$

• No trade (NT): A trader refrains from participating in the market which leads to zero profits at time t:  $\mathbb{E}\left(\Pi_{NT,t}^{R}|I_{t}\right) = 0.$ 

In case of equality of profits, we assume that a MO dominates LO and DO; and a LO dominates DO. If the expected profits of a MO are null, a rational trader refrains from trading.

Our model can be represented by a sequential game of incomplete information. The equilibrium definition used is as follows.

**Definition 1** A Perfect Bayesian Equilibrium (henceforth PBE) of the trading game is a strategy profile for all rational traders and belief system about other traders types at all information sets such that:

*i)* Sequential Rationality: Given the belief system, at each information set each trader's strategy specifies an optimal order that maximizes traders' expected profits given his beliefs and the strategies of other traders.

*ii)* Consistent beliefs: Given the strategy profile, the beliefs are consistent with Bayes rule (when appropriate).

In what follows, we focus on symmetric Perfect Bayesian Equilibria in pure strategies. A symmetric equilibrium refers to a situation where buyers and sellers with the same information (i.e, informed or uninformed) choose the same type of order in the first round of trading (except from the direction of trade).

#### 3 Equilibrium in the benchmark model without dark pool

We first consider the benchmark model without a dark pool (ND) where the available orders are:  $\mathbb{O}_{ND} = \{MO, LO, NT\}$ . The sequence of events can be seen in Figure (3). Note that the difference with respect to the timeline in Figure (1) when the dark pool is not available is that the liquidation value of the asset is revealed at t = 3.

We define  $\Omega_o$  and  $\Gamma_o$  as the probability that an informed trader and uninformed trader at t = 1 choose an order  $\mathcal{O} \in \mathbb{O}_{ND}$ , where o = 0 corresponds to a NT order; o = 1 to a MO; o = 2 to a LO;

t = 0	t = 1	t = 2	t = 3
Nature draws	• A trader may arrive	• A new trader may arrive	Liquidation value
$v \in \left\{ v^H, v^L \right\}$	• Rational traders choose their optimal submission strategies based on each trader's information set	• Rational traders choose their optimal submission strategies based on each trader's information set	is made public and the trading game is over
	• Liquidity traders trade according to their liquidity needs	• Liquidity traders trade according to their liquidity needs	

Figure 3: Time line of the trading game when traders do not have access to the dark pool

and such that  $\sum_{o=0}^{2} \Omega_o = 1$ , and  $\sum_{o=0}^{2} \Gamma_o = 1$ . We also define by  $\mathbb{B}$  the set of all possible states of the *LOB* at the end of the first trading period and by  $\mathcal{B}_1 \in \mathbb{B}$  a possible state of the book such that

 $\mathcal{B}_{1} = \begin{cases} \varnothing, & \text{if the best prices in the book are} \quad (A_{1}^{1}, B_{1}^{1}) \\ BMO, & \text{if the best prices in the book are} \quad (A_{1}^{2}, B_{1}^{1}) \\ BLO, & \text{if the best prices in the book are} \quad (A_{1}^{1}, B_{1}^{1} + \tau) \\ SMO, & \text{if the best prices in the book are} \quad (A_{1}^{1}, B_{1}^{2}) \\ SLO, & \text{if the best prices in the book are} \quad (A_{1}^{1} - \tau, B_{1}^{1}). \end{cases}$ 

Note that the state of the book  $\mathcal{B}_1 = \emptyset$ , can be obtained either because no trader arrived or a trader arrived but he decided not to trade, while the other states of the book are uniquely determined by the traders' actions at t = 1.

We solve the game backwards. At t = 2, the expected profits for an informed buyer and seller are summarized in Table 1, while Table 2 presents the expected profits of an uninformed buyer and seller.

	IH			IL		
	BMO	BLO	NT	SMO	SLO	NT
$(A_1^1, B_1^1)$	$(k_3 - k_1) \tau$	0	0	$(k_3 - k_1) \tau$	0	0
$(A_1^2, B_1^1)$	$(k_3-k_2) au$	0	0	$(k_3-k_1) au$	0	0
$(A_1^1, B_1^1 + \tau)$	$(k_3-k_1) au$	0	0	$(k_3 - k_1 + 1) \tau$	0	0
$(A_1^1, B_1^2)$	$(k_3-k_1) au$	0	0	$(k_3 - k_2) \tau$	0	0
$(A_1^1 - \tau, B_1^1)$	$(k_3 - k_1 + 1) \tau$	0	0	$(k_3 - k_1) \tau$	0	0

Table 1: Expected profits of an informed buyer (IH) and an informed seller (IL) at t = 2 when traders do not have access to the dark pool.

	UB			US		
	BMO	BLO	NT	SMO	SLO	NT
$(A_1^1, B_1^1)$	$-k_1 au$	0	0	$-k_1 au$	0	0
$(A_1^2, B_1^1)$	$(Xk_3-k_2)\tau$	0	0	$-(k_1+Xk_3)\tau$	0	0
$(A_1^1, B_1^1 + \tau)$	$(Yk_3-k_1)\tau$	0	0	$-\left(k_1-1+Yk_3\right)\tau$	0	0
$(A_1^1, B_1^2)$	$-\left(Xk_3+k_1\right)\tau$	0	0	$(Xk_3 - k_2)\tau$	0	0
$(A_1^1 - \tau, B_1^1)$	$-(Yk_3+k_1-1)\tau$	0	0	$(Yk_3 - k_1)\tau$	0	0

Table 2: Expected profits of an uninformed buyer (UB) and an uninformed seller (US) at t = 2 when traders do not have access to the dark pool.

Note that at t = 2 the expected profits of each strategy depend on the state of the LOB (which on its turn depends on the chosen strategy at t = 1). Uniformed traders at t = 2 form beliefs about the strategies and type of player in t = 1. Thus, we define the uninformed traders' belief at t = 2about the probability that the MO observed in the LOB was submitted by an informed trader as

$$X = \frac{\lambda \pi \Omega_1}{\eta + \lambda \pi \Omega_1 + \lambda \left(1 - \pi\right) \Gamma_1}.$$
(2)

Similarly, we define the uninformed traders' belief at t = 2 about the probability that the LO (observed in the LOB) was submitted by an informed trader as

$$Y = \frac{\pi \Omega_2}{\pi \Omega_2 + (1 - \pi) \Gamma_2}.$$
(3)

By comparing the expected profits of rational traders in t = 2 we obtain the following lemma.

**Lemma 1** In equilibrium the following results hold:

- at t = 2 an informed trader always submits a MO.
- at t = 2 an uninformed trader may submit either MO or NT, but never chooses LO. The optimal strategy for an uninformed trader is presented in Table 3:

An informed trader at t = 2 always chooses MO since it generates positive expected profits, while the expected profits of LO or NT are always null. An uninformed trader never chooses LOsince the probability of execution is 0, because no new orders arrive at t = 3 and, hence, the expected profits are null. An uninformed trader's choice at t = 2 depends on the state of the LOB since it reveals information. The choice will be either MO or NT. Without loss of generality, let us focus on an uninformed buyer at t = 2.<sup>9</sup> When the state of the LOB conveys no information

<sup>&</sup>lt;sup>9</sup>The argument for an uninformed seller follows since there exists the following symmetry: If the state of the LOB is  $(A_1^1, B_1^1)$ , uninformed sellers and buyers always make the same choice. If the state of the LOB is  $(A_1^2, B_1^1)$ , uninformed sellers choose the same as uninformed buyers when the LOB is  $(A_1^1, B_1^2)$ . If the state of the LOB is  $(A_1^1, B_1^1 + \tau)$  then uninformed sellers choose the same as uninformed buyers when the state of the LOB is  $(A_1^1, -\tau, B_1^1)$ .

State of the Book	UB	US
$(A_1^1, B_1^1)$	NT	NT
$(A_1^2, B_1^1)$	$\begin{cases} MO & \text{if } X > \frac{k_2}{k_3} \\ NT & \text{if } X \le \frac{k_2}{k_3} \end{cases}$	NT
$(A_1^1, B_1^1 + \tau)$	$\begin{cases} MO & \text{if } Y > \frac{k_1}{k_3} \\ NT & \text{if } Y \le \frac{k_1}{k_3} \end{cases}$	NT
$(A_1^1, B_1^2)$	NT	$\begin{cases} MO & \text{if } X > \frac{k_2}{k_3} \\ NT & \text{if } X \le \frac{k_2}{k_3} \end{cases}$
$(A_1^1 - \tau, B_1^1)$	NT	$\left\{ \begin{array}{ll} MO & \text{if } Y > \frac{k_1}{k_3} \\ NT & \text{if } Y \le \frac{k_1}{k_3} \end{array} \right.$

Table 3: Optimal trading strategies of an uniformed buyer (UB) and seller (US) at t = 2 when traders do not have access to the dark pool.

(i.e.,  $(A_1^1, B_1^1)$ ) then the optimal choice is NT since the expected profits of MO are negative. If the LOB reveals that at t = 1 there has been a BMO or BLO (i.e.,  $(A_1^2, B_1^1)$  or  $(A_1^1, B_1^1 + \tau)$ ), then the uninformed buyer at t = 2 chooses BMO if his belief that order came from an informed trader at t = 1 is sufficiently strong (i.e., X or Y is sufficiently large, respectively) so that expected profits are positive. Otherwise, the uninformed trader refrains from trading. In addition, if the state of the LOB is either  $(A_1^1, B_1^2)$  or  $(A_1^1 - \tau, B_1^1)$  then it reveals that the trader at t = 1 submitted a SMO or SLO, respectively, which implies that uninformed trader's expected profits of submitting a BMO at t = 2 are negative.

At t = 1, expected profits of an informed and uniformed trader are presented in Table 4 and Table 5, respectively.

IH	IL	Expected Profits
BMO	SMO	$\left(k_3-k_1 ight) au$
BLO	SLO	$\frac{\delta\eta}{2}\left(k_3+k_1-1\right)\tau$
NT	NT	0

Table 4: Expected profits of an informed buyer (IH) and seller (IL) at t = 1 when traders do not have access to the dark pool.

UB	US	$Expected \ Profits$
BMO	SMO	$-k_1 au$
BLO	SLO	$\frac{\delta}{2}\left(\left(\lambda\pi+\eta\right)\left(k_{1}-1\right)-\lambda\pi k_{3}\right)\tau$
NT	NT	0

Table 5: Expected profits of an uninformed buyer (UB) and seller (US) at t = 1 when traders do not have access to the dark pool.

The following lemma presents the informed and uninformed traders' optimal strategies at t = 1.

Lemma 2 In equilibrium the following results hold:

- at t = 1 an informed trader never chooses NT.
- at t = 1 an uniformed trader never chooses a MO.

At t = 1, an informed trader never chooses NT since it is always dominated by at least a MO and, hence, an informed trader may either choose MO or LO. In contrast, an uninformed trader at t = 1 never chooses a MO since its expected profits are negative, and it is always dominated by at least the NT strategy and, consequently, an uninformed trader at t = 1 may either choose LO or NT. Hence, the candidate strategy profiles at t = 1 that can be sustained as a symmetric PBE are:

$$(BMO, SMO, BLO, SLO), (BMO, SMO, NT, NT), \\ (BLO, SLO, BLO, BLO), (BLO, SLO, NT, NT),$$

where the two first components correspond to strategies of informed traders at t = 1 (*IH* and *IL*, respectively) and the two last components correspond to strategies of uninformed traders at t = 1 (*UB* and *US*, respectively).

We are now in a position to characterize the PBE of the reduced trading game where the dark pool is not available.

**Proposition 1** If  $k_1 > 1$ , then a PBE of the game is as follows:

•  $\mathcal{E}_1^{ND}$ : (BMO, SMO, BLO, SLO) is the optimal strategy profile at t = 1 if

Conditions
$k_3 - k_1 \ge \delta \frac{\eta}{2} \left( k_3 + k_1 - 1 \right)$ and
$\left(\lambda\pi+\eta\right)\left(k_1-1\right)-\lambda\pi k_3>0.$

The beliefs of an uninformed trader at t = 2 are:  $X = \frac{\lambda \pi}{\eta + \lambda \pi}$  and Y = 0. The optimal strategy of an informed trader at t = 2 is to choose MO for all possible states of the LOB, and

the optimal strategy of an uninformed trader at t = 2 is to choose NT for all possible states of the LOB.

•  $\mathcal{E}_2^{ND}$ : (BMO, SMO, NT, NT) is the optimal strategy profile at t = 1 if

Conditions	-
$k_3 - k_1 \ge \delta \frac{\eta}{2} \left( k_3 + k_1 - 1 \right)$ and	
$0 \ge \left(\lambda \pi + \eta\right) \left(k_1 - 1\right) - \lambda \pi k_3.$	

The beliefs of an uninformed trader at t = 2 are:  $X = \frac{\lambda \pi}{\eta + \lambda \pi}$  and  $Y = p \in [0, 1]$ . The optimal strategy of an informed trader at t = 2 is to choose MO for all possible states of the LOB, and the optimal strategy of an uninformed trader at t = 2 is described in Table A.2 of Appendix A.

•  $\mathcal{E}_3^{ND}$ : (BLO, SLO, BLO, BLO) is the optimal strategy profile at t = 1 if

Conditions
$\delta \frac{\eta}{2} (k_3 + k_1 - 1) > k_3 - k_1 \text{ and}$
$(\lambda \pi + \eta) \left(k_1 - 1\right) - \lambda \pi k_3 > 0.$

The beliefs of an uninformed trader at t = 2 are: X = 0 and  $Y = \pi$ . The optimal strategy of an informed trader at t = 2 is to choose MO for all possible states of the LOB, and the optimal strategy of an uninformed trader at t = 2 is described in Table A.3 of Appendix A.

•  $\mathcal{E}_4^{ND}$ : (BLO, SLO, NT, NT) is the optimal strategy profile at t = 1 if

Conditions
$\delta \frac{\eta}{2} (k_3 + k_1 - 1) > k_3 - k_1 \text{ and}$
$0 \ge (\lambda \pi + \eta) \left( k_1 - 1 \right) - \lambda \pi k_3.$

The beliefs of an uninformed trader at t = 2 are: X = 0 and Y = 1. The optimal strategy of an informed trader at t = 2 is to choose MO for all possible states of the LOB, and the optimal strategy of an uninformed trader at t = 2 is described in Table A.4 of Appendix A.<sup>10</sup>

Some of the features of Proposition 1 are illustrated in Figure 4. The left panel shows the optimal strategy of a trader that arrives at t = 1 when there is no access to the dark pool as a function of the probability that a liquidity trader arrives,  $\eta$ , and the discount factor,  $\delta$ , if there is low adverse selection (measured as the probability of arrival of an informed trader). The right panel presents the case of high adverse selection. In addition, the next corollary summarizes under which conditions informed and uninformed traders at t = 1 find an order type relatively more attractive than the alternative, keeping the other factors fixed.

<sup>&</sup>lt;sup>10</sup>Proposition 4 in Appendix A characterizes the PBE when  $k_1 = 1$ .



Figure 4: Optimal strategies at t = 1 without dark pool. Parameters values:  $k_1 = 2, k_2 = 3, k_3 = 4, \lambda = 0.5$ . Left Panel  $\pi = 0.15$ , Right Panel  $\pi = 0.95$ .

**Corollary 1** At t = 1, when there is no access to the dark pool:

- An informed trader finds that expected profits of a MO increase in relation to those of a LO when σ increases, when k<sub>1</sub>, δ, η or τ decreases and the choice between a MO or LO does not depend on λ nor π.
- An uninformed trader finds that expected profits of a LO increase in relation to those of a NT when  $k_1$ ,  $\eta$ ,  $\delta$  or  $\tau$  increases and when  $\pi$  or  $\lambda$  or  $\sigma$  decreases.

The optimal strategy profiles at t = 1 depend on trader characteristics ( $\delta$ ) and stock market characteristics ( $k_1$ ,  $\sigma$ ,  $\pi$ ,  $\tau$ ,  $\lambda$ ,  $\eta$ ). Notice that we can understand  $1/k_1$  as a measure of liquidity;  $\sigma$  as a measure of the asset's volatility; and the probability that the trader is informed,  $\pi$ , as a measure of adverse selection. The optimal strategy profiles at t = 1 described in Proposition 1 are the result of combining how trader ( $\delta$ ) and stock market characteristics ( $k_1$ ,  $\sigma$ ,  $\tau$ ,  $\lambda$ ,  $\pi$ ,  $\eta$ ) impact the optimal order choice for an informed and an uninformed trader. These are described below.

Let us discuss how trader characteristics impact the optimal choice of strategy profile at t = 1. Informed traders choose between MO and LO. If the probability that a liquidity trader arrives is not too large, then an informed trader at t = 1 chooses a MO since execution is guaranteed, otherwise he chooses a LO. This is because, as  $\eta$  increases the relative attractiveness of a LO of an informed trader raises since there is a higher probability of execution. Uninformed traders select between NT and participating with LO. When adverse selection is high, the uninformed trader chooses NT: if the LO is executed due to a MO of the opposite sign submitted by informed trader at t = 2, then it reveals that the value of the asset is low (if a BLO has been submitted) and high (if a SLO has been submitted). Otherwise, the uninformed trader chooses LO. In contrast, notice that the profits of an informed trader do not depend on the probability that an informed trader arrives  $(\lambda \pi)$  in the next trading period. This is because the probability of execution of a LO submitted by an informed trader only depends on the probability that a liquidity trader arrives in the next trading period, and that the liquidity trader submits a MO of a different sign with respect to the initial LO.<sup>11</sup> Furthermore, notice that a LO is attractive when the discount factor is high since traders do not give a high value to immediacy.

In terms of stock market characteristics, low liquidity, high volatility or low tick size (high  $k_1$ , high  $\sigma$ , low  $\tau$  respectively) foster that LO are less attractive since the potential increment in profits does not compensate for the execution risk. Hence, if the asset has low liquidity, high volatility or low tick size, then an uninformed trader prefers NT to LO, and an informed trader prefers MO to LO.

### 4 Equilibrium in a model with dark pool

We next consider a model where rational traders have both access to the exchange and to the dark pool and can submit the orders in (1). We define  $\Omega_3$  and  $\Gamma_3$  as the probability that an informed trader and uninformed trader at t = 1 choose a DO, and such that  $\sum_{o=0}^{3} \Omega_o = 1$ , and  $\sum_{o=0}^{3} \Gamma_o = 1$ . Note that the set of the possible states of the book is the same as in the case there is no dark pool but the state of the book  $\mathcal{B}_1 = \emptyset$ , can be obtained in this case either because no trader arrived or a trader arrived and decided not to trade or because a trader arrived and he submitted a DO.

We solve the model backwards. At t = 2 the expected profits of each strategy depend on the state of the LOB. Additionally, uninformed traders form beliefs about the strategies that have been chosen at t = 1. Let X and Y be defined as in (2) and (3), respectively, and Z denote the uninformed trader's belief at t = 2 about the probability that a DO was submitted by an informed, which is equal to

$$Z = \frac{\pi\Omega_3}{\pi\Omega_3 + (1-\pi)\,\Gamma_3}.\tag{4}$$

As in the case when the dark pool was not available, and without loss of generality, we will focus on the expected profits for an informed buyer at t = 2, as summarized in Table 6 below.

<sup>&</sup>lt;sup>11</sup>An informed trader that submits a LO at t = 1 knows that the LO will not be executed in the next trading period against an order submitted by an informed trader since an informed trader at t = 2 chooses an order of the same sign as the initial order. In addition, an informed trader at t = 1 correctly predicts that an uninformed trader at t = 2 never submits a MO of the opposite sign as the informed trader at t = 1.

IH	BMO	BDO	BLO	NT
$(A_1^1, B_1^1)$	$(k_3 - k_1) \tau$	$ heta k_3  au$	$P_I\delta\left(k_1+k_3-1\right)\tau$	0
$(A_1^2, B_1^1)$	$(k_3-k_2) au$	$\theta\left(k_3 - \frac{k_2 - k_1}{2}\right)\tau$	0	0
$(A_1^1, B_1^1 + \tau)$	$(k_3-k_1)\tau$	$\theta\left(k_3-\frac{1}{2}\right) au$	0	0
$(A_1^1, B_1^2)$	$\left(k_3-k_1\right)\tau$	$\theta\left(k_3 + \frac{k_2 - k_1}{2}\right)\tau$	0	0
$(A_1^1 - \tau, B_1^1)$	$(k_3 - k_1 + 1)\tau$	$\theta\left(k_3+\frac{1}{2}\right) au$	0	0

Table 6: Expected profits of an informed buyer (IH) at t = 2

where  $P_I$  is the probability of execution of a limit order placed by an informed trader at t = 2 conditional on the fact that there is no change in the LOB during the first trading period, and equals

$$P_{I} = p_{BLO,2}^{IH} \left( \mathcal{B}_{1} = \varnothing \right) = \frac{\lambda (1-\theta) \frac{1-\pi}{2} \Gamma_{3}}{1-\lambda - \eta + \lambda \left( \pi \Omega_{3} + (1-\pi) (\Gamma_{0} + \Gamma_{3}) \right)}$$

Similarly, the expected profits of an uninformed buyer at t = 2 are summarized in Table 7.

UB	BMO	BDO	BLO	NT
$(A_1^1, B_1^1)$	$-k_1 au$	0	$P_U\delta\left(k_1 - Zk_3 - 1\right)\tau$	0
$(A_1^2, B_1^1)$	$(Xk_3-k_2)\tau$	$\theta\left(Xk_3 - \frac{k_2 - k_1}{2}\right) au$	0	0
$(A_1^1, B_1^1 + \tau)$	$(Yk_3 - k_1)\tau$	$\theta\left(Yk_3-\frac{1}{2}\right) au$	0	0
$(A_1^1, B_1^2)$	$-\left(Xk_3+k_1\right)\tau$	$-\theta\left(Xk_3-\frac{k_2-k_1}{2}\right) au$	0	0
$(A_1^1 - \tau, B_1^1)$	$-(Yk_3+k_1-1)\tau$	$-\theta\left(Yk_3-\frac{1}{2}\right) au$	0	0

Table 7: Expected profits of an uninformed buyer (UB) at t = 2

where  $P_U$  is the probability of execution of a limit order placed by an uninformed trader at t = 2 given that there are no changes in prices in the LOB during the first trading period, and equals

$$P_U = p_{BLO,2}^{UB} \left( \mathcal{B}_1 = \varnothing \right) = \frac{\frac{1}{2}\lambda(1-\theta)(\pi\Omega_3 + (1-\pi)\Gamma_3)}{1-\lambda - \eta + \lambda(\pi\Omega_3 + (1-\pi)(\Gamma_0 + \Gamma_3))}.$$

At t = 1 the expected profits of an informed IH and an uniformed buyer UB are summarized in Table 8 and Table 9, respectively.<sup>12</sup>

Notice that the expected profits of a BDO submitted by an uninformed trader at t = 1 can be rewritten as

$$\theta \cdot 0 + \max\left\{ (1-\theta)\delta^2 \left( -k_1\tau + \frac{\lambda}{2} \left( \pi I_{SLO,2}^{I,L,\mathcal{B}_1 = \emptyset} + (1-\pi) I_{SLO,2}^{U,S,\mathcal{B}_1 = \emptyset} \right) \tau - \left( \frac{\lambda\pi}{2} I_{BMO,2}^{I,H,\mathcal{B}_1 = \emptyset} + \frac{\eta}{2} \right) (k_2 - k_1) \tau \right\}, 0 \right\}.$$

This expression indicates that if a *BDO* gets executed at t = 1, then its expected profits are zero.

<sup>&</sup>lt;sup>12</sup>Notice that due to the symmetry of the game, the expected profits of the informed IL trader and uninformed seller US are the same as the ones displayed in Tables 8 and 9, respectively.

IH	Expected Profits at $t = 1$
BMO	$\left(k_3-k_1 ight) au$
BLO	$rac{\eta\delta}{2}\left(k_{3}+k_{1}-1 ight) au$
BDO	$\theta k_3 \tau + \max\left\{ (1-\theta)\delta^2 \left( \lambda \frac{(1-\pi)}{2} I_{SLO,2}^{US,\mathcal{B}_1=\varnothing} + (k_3-k_1) - (k_2-k_1) \left( \lambda \pi I_{BMO,2}^{IH,\mathcal{B}_1=\varnothing} + \frac{\eta}{2} \right) \right) \tau, 0 \right\}$
NT	0

Table 8: Expected profits of an informed buyer (IH) at t = 1

UB	Expected Profits at $t = 1$
BMO	$-k_1 au$
BLO	$\frac{\delta}{2} \left( \eta(k_1 - 1) - \lambda \pi I_{SMO,2}^{IL,\mathcal{B}_1 = BLO}(k_3 - k_1 - 1) \right) \tau$
BDO	$\max\left\{ (1-\theta)\delta^2 \left( \frac{\lambda}{2} (\pi I_{SLO,2}^{IL,\mathcal{B}_1=\varnothing} + (1-\pi) I_{SLO,2}^{US,\mathcal{B}_1=\varnothing}) + \right. \\ \left. + \left( \frac{\lambda\pi}{2} I_{BMO,2}^{IH,\mathcal{B}_1=\varnothing} + \frac{\eta}{2} \right) (k_1 - k_2) - k_1 \right) \tau, 0 \right\}$
NT	0

Table 9: Expected profits of an uninformed buyer (UB) at t = 1

Otherwise (if the order is not attended executed at t = 1 and returns to the market at t = 3), the expected profits are given by the second summand of the previous formula. The expected profits depend on whether a trader who returns to the market decides to submit a MO or NT. If he submits a MO then the profit consists of the product of the probability of no execution in the dark pool (i.e.,  $1-\theta$ ) and the squared discount factor (i.e.,  $\delta^2$ ) and by the expected profits of a BMO at t=1 for an UB (i.e.,  $-k_1\tau$ ) adjusted by two terms. The first one,  $\frac{\lambda}{2} \left(\pi I_{SLO,2}^{IL,\mathcal{B}_1=\varnothing} + (1-\pi)I_{SLO,2}^{US,\mathcal{B}_1=\varnothing}\right)\tau$ , shows the increase in the expected profits a UB due to the possibility that at t = 2 a new trader arrives and submits a SLO. The second one,  $-\left(\frac{\lambda}{2}\pi I_{BMO,2}^{IH,\mathcal{B}_1=\varnothing} + \frac{\eta}{2}\right)(k_2 - k_1)\tau$ , shows the decrease in expected profits due to the possibility that a trader at t = 2 submits a BMO and, consequently, the MO that arrives at t = 3 from the dark pool is executed at a worse price. Notice that the increase in expected profits due to the potential arrival of a SLO at t = 2 does not compensate for the negative profits of a MO and, as we have point out, these losses might be even greater in case that a BMO is submitted at t = 2. If he selects NT then the expected profits equal zero.

By comparing the expected profits of each of the possible strategies for each type of rational trader at t = 2, Lemma 3 states the strategies that are dominated and, hence, never chosen by a rational player.

#### Lemma 3 In equilibrium the following results hold:

• at t = 2 an informed trader never chooses a LO or NT. The optimal strategy depends on the value of the parameters as explained in Table B.1 in Appendix B.

• at t = 2 an uninformed trader at t = 2 never chooses a LO. The optimal strategy depends on the value of the parameters as explained in Table B.2 in Appendix B.

An informed trader at t = 2 may choose between MO or DO, and never chooses NT or LO since NT is always dominated by MO, and LO is never executed since: a) if the LOB has changed, then no MO can arrive at t = 3; b) if the LOB has not changed, then BLO (SLO) can only be executed if an uninformed seller (buyer) at t = 1 chooses DO, but as we have discussed this cannot occur in equilibrium.

An uninformed trader at t = 2 might choose between MO, DO or NT order, but never LO since: a) if the LOB has changed, then no MO arrives at t = 3 and, hence, it has zero probability of execution; b) if the LOB has not changed, then the LO can only be executed if a trader at t = 1 has chosen a DO. However, as we have explained after Table 9, we know that it is never optimal for an uninformed trader at t = 1 to choose a DO. Hence the trader at t = 2 forms the correct beliefs that, if a LO is executed at t = 3, it must have come from an informed trader at t = 1 with probability 1. But this information reveals to the uninformed buyer (seller) that the value of the asset must be low (high) and, hence, expected profits of a LO are negative.

By comparing the expected profits for each type of orders at t = 1, we find the optimal strategies chosen by informed and uninformed traders.

#### Lemma 4 In equilibrium the following results hold:

- at t = 1 an informed trader never chooses NT.
- at t = 1 an uniformed trader never chooses a MO or a DO.

An informed trader at t = 1 chooses among MO, LO or DO, but never NT since it is always dominated by at least a MO. An uninformed trader at t = 1 may choose between LO or NT since the expected profits of a MO are negative, and also the expected profits of a DO as explained after Table 9.

Hence, the candidate strategy profiles at t = 1 that can be sustained as a *PBE* are:

$$(BMO, SMO, BLO, SLO), (BMO, SMO, NT, NT), (BLO, SLO, BLO, BLO), (BLO, SLO, NT, NT), (BDO, SDO, BLO, SLO), (BDO, SDO, NT, NT),$$

where, as before, the two first components correspond to strategies of informed traders at t = 1 (*IH* and *IL*, respectively) and the two last components correspond to strategies of uninformed traders at t = 1 (*UB* and *US*, respectively).

The PBE of the trading game where rational traders have access to a dark pool is characterized as follows.

**Proposition 2** If  $k_1 > 1$ , then a PBE of the game is as follows:

•  $\mathcal{E}_1^D$ : (BMO, SMO, BLO, SLO) is the optimal strategy profile at t = 1 if

Conditions				
	$k_3 - k_1 \ge \frac{\eta}{2} \delta \left( k_3 + k_1 - 1 \right),$			
$\theta \le \frac{k_3 - k_1}{k_3}$	$k_3 - k_1 \ge \theta k_3 + (1 - \theta) \delta^2 \left( k_3 - k_1 - (k_2 - k_1) \left( \lambda \pi + \frac{\eta}{2} \right) \right), and$			
	$\left(\lambda\pi+\eta\right)\left(k_1-1\right)-\lambda\pi k_3>0.$			

The beliefs of an uninformed trader at t = 2 are:  $X = \frac{\lambda \pi}{\eta + \lambda \pi}$ , Y = 0 and  $Z = q \in [0, 1]$ . The optimal strategy of an informed trader and an uninformed at t = 2 are described in Tables B.1 and B.5 of Appendix B, respectively.

•  $\mathcal{E}_2^D$ : (BMO, SMO, NT, NT) is the optimal strategy profile at t = 1 if

	Conditions
	$k_3 - k_1 \ge \frac{\eta}{2} \delta \left( k_3 + k_1 - 1 \right),$
$\theta \le \frac{k_3 - k_1}{k_3}$	$k_3 - k_1 \ge \theta k_3 + (1 - \theta) \delta^2 \left( (k_3 - k_1) - (k_2 - k_1) \left( \lambda \pi + \frac{\eta}{2} \right) \right), \text{ and}$
	$0 \ge (\lambda \pi + \eta) \left( k_1 - 1 \right) - \lambda \pi k_3.$

The beliefs of an uninformed trader at t = 2 are:  $X = \frac{\lambda \pi}{\eta + \lambda \pi}$ ,  $Y = p \in [0, 1]$  and  $Z = q \in [0, 1]$ . The optimal strategy of an informed trader and an uninformed at t = 2 are described in Tables B.1 and B.6 of Appendix B, respectively.

•  $\mathcal{E}_3^D$ : (BLO, SLO, BLO, BLO) is the optimal strategy profile at t = 1 if

Conditions			
	$\frac{\eta}{2}\delta(k_3 + k_1 - 1) \ge \theta k_3 + (1 - \theta)\delta^2\left((k_3 - k_1) - (k_2 - k_1)\left(\lambda\pi + \frac{\eta}{2}\right)\right),$		
$ heta \le rac{k_3-k_1}{k_3}$	$\frac{\eta}{2}\delta(k_3+k_1-1) > k_3-k_1, and$		
	$(\lambda \pi + \eta) \left(k_1 - 1\right) - \lambda \pi k_3 > 0.$		
$k_3 - k_1 \neq 0 \neq k_3 - k_1 + 1$	$\frac{\eta}{2}\delta(k_3 + k_1 - 1) \ge \theta k_3 + (1 - \theta)\delta^2\left((k_3 - k_1) - (k_2 - k_1)\left(\frac{\eta}{2}\right)\right)  and$		
$k_3  \langle v \leq k_3 + \frac{1}{2}$	$(\lambda \pi + \eta) \left(k_1 - 1\right) - \lambda \pi k_3 > 0.$		
$\frac{k_3 - k_1 + 1}{k_3 + \frac{1}{2}} < \theta$	$\frac{\eta}{2}\delta(k_3 + k_1 - 1) \ge \theta k_3 + (1 - \theta)\delta^2\left((k_3 - k_1) - (k_2 - k_1)\left(\frac{\eta}{2}\right)\right).$		

The beliefs of an uninformed trader at t = 2 are: X = 0,  $Y = \pi$  and  $Z = q \in [0, 1]$ . The optimal strategy of an informed trader and an uninformed at t = 2 are described in Tables B.1 and B.7 of Appendix B, respectively.

Conditions			
	$\frac{\eta}{2}\delta(k_3 + k_1 - 1) \ge \theta k_3 + (1 - \theta)\delta^2\left((k_3 - k_1) - (k_2 - k_1)\left(\lambda\pi + \frac{\eta}{2}\right)\right),$		
$ heta \le rac{k_3 - k_1}{k_3}$	$\frac{\eta}{2}\delta(k_3+k_1-1) > (k_3-k_1), and$		
	$0 \ge (\lambda \pi + \eta) \left(k_1 - 1\right) - \lambda \pi k_3.$		
$\frac{\frac{k_3 - k_1}{k_3} < \theta \le \frac{k_3 - k_1 + 1}{k_3 + \frac{1}{2}}}{$	$\frac{\eta}{2}\delta(k_3 + k_1 - 1) \ge \theta k_3 + (1 - \theta)\delta^2\left((k_3 - k_1) - (k_2 - k_1)\left(\frac{\eta}{2}\right)\right) and$		
	$0 \ge (\lambda \pi + \eta) \left(k_1 - 1\right) - \lambda \pi k_3.$		

•  $\mathcal{E}_4^D$ : (BLO, SLO, NT, NT) is the optimal strategy profile of a trader at t = 1 if

The beliefs of an uninformed trader at t = 2 are: X = 0, Y = 1 and  $Z = q \in [0, 1]$ . The optimal strategy of an informed trader and an uninformed at t = 2 are described in Tables B.1 and B.8 of Appendix B, respectively.

•  $\mathcal{E}_5^D$ : (BDO, SDO, BLO, SLO) is the optimal strategy profile of a trader at t = 1 if

Conditions			
	$\theta k_3 + (1-\theta)\delta^2 \left(k_3 - k_1 - (k_2 - k_1)\left(\lambda \pi + \frac{\eta}{2}\right)\right) > k_3 - k_1,$		
$ heta \le rac{k_3-k_1}{k_3}$	$\theta k_3 + (1-\theta)\delta^2 \left(k_3 - k_1 - (k_2 - k_1)\left(\lambda \pi + \frac{\eta}{2}\right)\right) > \frac{\eta}{2}\delta \left(k_3 + k_1 - 1\right), \text{ and}$		
	$(\lambda \pi + \eta)(k_1 - 1) - \lambda \pi k_3 > 0.$		
$k_3 - k_1 < \theta < k_3 - k_1 + 1$	$\theta k_3 + (1-\theta)\delta^2 \left(k_3 - k_1 - (k_2 - k_1)\left(\frac{\eta}{2}\right)\right) > \frac{\eta}{2}\delta \left(k_3 + k_1 - 1\right)$ and		
$k_3  \langle v \leq -k_3 + \frac{1}{2}$	$((\lambda \pi + \eta)(k_1 - 1) - \lambda \pi k_3) > 0.$		
$\frac{k_3 - k_1 + 1}{k_3 + \frac{1}{2}} <  heta$	$\theta k_3 + (1-\theta)\delta^2 \left(k_3 - k_1 - (k_2 - k_1)\left(\frac{\eta}{2}\right)\right) > \frac{\eta}{2}\delta \left(k_3 + k_1 - 1\right).$		

The beliefs of an uninformed trader at t = 2 are: X = 0, Y = 0 and Z = 1. The optimal strategy of an informed trader and an uninformed at t = 2 are described in Tables B.1 and B.9 of Appendix B, respectively.

•  $\mathcal{E}_6^D$ : (BDO, SDO, NT, NT) is the optimal strategy profile of a trader at t = 1 if

Conditions			
	$\theta k_3 + (1-\theta)\delta^2 \left( (k_3 - k_1) - (k_2 - k_1) \left( \lambda \pi + \frac{\eta}{2} \right) \right) > (k_3 - k_1),$		
$ heta \leq rac{k_3-k_1}{k_3}$	$\theta k_3 + (1-\theta)\delta^2 \left( (k_3 - k_1) - (k_2 - k_1) \left( \lambda \pi + \frac{\eta}{2} \right) \right) > \frac{\eta}{2} \delta \left( k_3 + k_1 - 1 \right),$		
	$0 \ge (\lambda \pi + \eta) \left( k_1 - 1 \right) - \lambda \pi k_3.$		
$\frac{k_3 - k_1}{k_3} < \theta \le \frac{k_3 - k_1 + 1}{k_3 + \frac{1}{2}}$	$\theta k_3 + (1-\theta)\delta^2 \left( (k_3 - k_1) - (k_2 - k_1) \left(\frac{\eta}{2}\right) \right) > \frac{\eta}{2}\delta \left( k_3 + k_1 - 1 \right),$		
	$0 \ge (\lambda \pi + \eta) \left( k_1 - 1 \right) - \lambda \pi k_3.$		

The beliefs of an uninformed trader at t = 2 are: X = 0,  $Y = p \in [0,1]$  and Z = 1. The optimal strategy of an informed trader and an uninformed at t = 2 are described in Tables B.1 and B.10 of Appendix B, respectively.<sup>13</sup>

<sup>&</sup>lt;sup>13</sup>Proposition 5 in Appendix B characterizes the PBE when  $k_1 = 1$ .

Proposition 2 shows that having access to a dark pool changes the optimal submission strategy profiles at t = 1 for informed and uninformed traders. In addition, Proposition 2 (and Appendix B) characterize the optimal trading strategies for informed and uninformed traders at t = 2. Without loss of generality, let us focus on the optimal strategy of a buyer at t = 2. Informed traders at t = 2 submit BMO for all states of the LOB when the execution risk in the dark is high  $(\theta \leq \frac{k_3-k_2}{k_3-\frac{k_2-k_1}{2}})$  and submit BDO for all the states of the LOB when the execution risk in the dark is  $(\theta \geq \frac{k_3-k_1+1}{k_3+\frac{1}{2}})$ . As the execution risk in the dark lowers, informed traders replace BMO by BDO in the following order of states of the LOB:  $(A_1^2, B_1^1), (A_1^1, B_1^2), (A_1^1, B_1^1), (A_1^1, B_1^1 + \tau), (A_1^1 - \tau, B_1^1)$ . This is because, when a BMO had been submitted at t = 1 (state of the LOB :  $(A_1^2, -\tau, B_1^1)$ ), then the gain from another BMO is small in relation to a BDO for high execution risk, while when a state of the BMO is large in relation to a BDO for low execution risk.

The optimal strategy of an uninformed trader at t = 2 critically depends on the set of the uninformed trader's beliefs at t = 2, X, Y, Z, about the probability that a MO, LO, DO order, respectively, was submitted by an informed trader at t = 1. These beliefs are different given the optimal strategy profile at t = 1. When the state of the LOB contains no information, i.e.,  $(A_1^1, B_1^1)$ , then uninformed traders at t = 2 submit NT orders since the expected profits of a MO are negative, and the profits of a DO are zero because the mid-point price is equal to the unconditional expected value of the asset,  $\mu$ . When the state of the LOB indicates that a BMO (SMO) had been submitted at t = 1, with state of the  $LOB : (A_1^2, B_1^1) ((A_1^1, B_1^2))$  and it is optimal that informed traders submit this strategy profile (i.e., in  $\mathcal{E}_1^D$  and  $\mathcal{E}_2^D$ ), then uniformed buyers at t = 2 may submit BMO, BDOor NT (BDO or NT) depending on which of the expected profits is higher. In all other possible equilibrium strategy profiles, uninformed buyers prefer NT (BDO). When the state of the LOBindicates that a BLO (SLO) had been submitted at t = 1, with state of the  $LOB : (A_1^1, B_1^1 + \tau)$  $((A_1^1 - \tau, B_1^1))$ , and it is optimal that uninformed traders submit this strategy profile (i.e., in  $\mathcal{E}_1^D$ and  $\mathcal{E}_5^D$ ), then uniformed buyers at t = 2 prefer NT (BDO). Otherwise, uninformed traders prefer BMO, BDO or NT (BDO or NT) depending on which of the expected profits is higher.

Figures 5 and 6 show the optimal strategy of a trader that arrives at t = 1 when a dark pool is available for different levels of execution risk in the dark pool (lowest in the upper right graph and highest in the lower right graph) as a function of the discount factor and the probability that a liquidity trader arrives for markets with low and high adverse selection, respectively.

We observe that when the execution risk is high (low  $\theta$ ) the optimal strategy of the uninformed trader does not change in relation to when the dark pool is unavailable. Adding a dark pool alongside the exchange also changes the optimal strategy of an uninformed trader at t = 1 even if an uninformed trader never goes to the dark. When the execution risk in the dark is low, uninformed traders may switch from NT to using LO in the exchange. This is because the mere existence of the dark market offers the possibility for informed traders at t = 2 to migrate from the exchange to the dark and, consequently, adverse selection is reduced in the exchange. This induces uninformed



Figure 5: Optimal strategies at t = 1 with dark pool as a function of the probability that a liquidity trader arrives,  $\eta$ , and the discount factor,  $\delta$ . Parameter values:  $k_1 = 5$ ,  $k_2 = 6$ ,  $k_3 = 7$ ,  $\lambda = 0.5$ ,  $\pi = 0.15$ . Values for the execution probability in the dark,  $\theta$ , are specified above each graph.

traders to switch from NT to LO. An informed trader's optimal strategy at t = 1 changes when the dark pool is available: as the probability of execution in the dark increases, informed traders gradually replace MO by DO, and as  $\theta$  is even larger they also replace LO by DO.

Proposition 2 indicates that the conditions for a PBE to exist depend on how the execution risk in the dark pool,  $\theta$ , compares to two cutoffs related to the asset's volatility  $\sigma$  and liquidity  $(1/k_1)$ since these influence the prices of the LOB and the possible realizations of the liquidation value. Specifically, strategy profiles where an informed trader at t = 1 submits a MO cannot be part of a PBE if  $\frac{k_3 - k_1}{k_3} < \theta$  since these strategies are dominated by LO or DO because prices are more attractive given their execution risk. Additionally, strategy profiles where an uninformed trader at t = 1 chooses NT cannot be equilibrium strategy profiles at t = 1 if  $\frac{k_3 - k_1 + 1}{k_3 + \frac{1}{2}} < \theta$ . Hence, if the execution probability in the dark,  $\theta$ , changes continuously, then optimal strategy profiles at t = 1may present discontinuities. Second, even though optimal strategies profiles are unique at t = 1



Figure 6: Optimal strategies at t = 1 with dark pool as a function of the probability that a liquidity trader arrives,  $\eta$ , and the discount factor,  $\delta$ . Parameter values:  $k_1 = 5$ ,  $k_2 = 6$ ,  $k_3 = 7$ ,  $\lambda = 0.5$ ,  $\pi = 0.95$ . Values for the execution probability in the dark,  $\theta$ , are specified above each graph.

for given parameter values, multiple equilibria may exist except when (BDO, SDO, BLO, SLO)is optimal. Multiple equilibria exist since there is a continuum of uninformed trader's beliefs at t = 2 that can sustain the *PBE*. However, when (BDO, SDO, BLO, SLO) is optimal at t = 1, the beliefs of uninformed traders at t = 2 are uniquely determined since X = 0, Y = 0 and Z = 1.

We analyze which PBE exists in a world with almost no adverse selection (i.e., if  $\pi$  is low enough).

**Corollary 2** Suppose that  $k_1 > 1$  and that  $\pi$  is low enough. At t = 1, the strategy profiles that are part of the PBE are the following: (BMO, SMO, BLO, SLO), (BLO, SLO, BLO, SLO), and (BDO, SDO, BLO, SLO).

The intuition is as follows. When the probability that an informed trader arrives is low enough, uninformed traders realize that they are very likely to trade with liquidity traders instead of informed traders and have higher expected profits with LO than with NT. In contrast, when  $\pi$  is high, then adverse selection is high because if the LO is executed it is very likely that it is because a MO of the opposite sign has been submitted by an informed trader. Then this reveals that the value of the asset is low (if a BLO was submitted) and high (if a SLO was submitted). Hence, due to adverse selection, uninformed traders do not participate in the market since the probability of obtaining negative profits is higher.

#### 5 Market quality and welfare analysis

In the presence of asymmetric information, we study how market quality and welfare are affected by the dark pool. Hence, we compare several measures of market quality and welfare when the dark pool is available (Section 4) to when it is unavailable (Section 3). The measures of market quality that we consider are: expected inside spread, expected traded volume, and expected price informativeness.

We denote by  $\mathcal{E}_i^a$  an equilibrium, where a = D, ND indicates whether we consider that the dark pool (D) is available or unavailable (ND). Note that if the dark pool is available we have six equilibria  $\mathcal{E}_i^D$  with i = 1, ..., 6, while if the dark pool is not available we have four equilibria  $\mathcal{E}_i^{ND}$  for i = 1, ..., 4.

We proceed as follows for market quality and welfare at t = 1. First, we compute the market quality and welfare measures for each of the possible strategy profiles when the dark pool is not available,  $\mathcal{E}_i^{ND}$  for i = 1, ..., 4. Second, when traders have access to the dark pool, we assume that  $\theta$ is uniformly distributed in the interval [0, 1]. Then, starting from each of the four possible equilibria when traders do not have access to the dark pool ( $\mathcal{E}_i^{ND}$  for i = 1, ..., 4), we find the optimal strategy profile at t = 1 when traders have access to the dark pool for all the values of  $\theta$ . For example, starting from strategy (*BMO*, *SMO*, *NT*, *NT*), we find the following cutoffs of  $\theta$  for which traders switch their optimal trading strategy when the dark pool is available. We define first,

$$\overline{\theta}_{1} \equiv \frac{k_{3} - k_{1} - \delta^{2} \left(k_{3} - k_{1} + \left(k_{1} - k_{2}\right) \left(\frac{1}{2}\eta + \pi\lambda\right)\right)}{\left(k_{3} - \delta^{2} \left(k_{3} - k_{1} + \left(k_{1} - k_{2}\right) \left(\frac{1}{2}\eta + \pi\lambda\right)\right)\right)},$$

and write therefore:

$Optimal \ Trading \ Strategy \ at \ t = 1 \ starting \ from \ (BMO, SMO, NT, NT)$			
(BMO, SMO, NT, NT)	if	$ heta \leq \overline{ heta}_1$	
(BDO, SDO, NT, NT)	if	$\overline{\theta}_1 < \theta \le \frac{k_3 - k_1 + 1}{k_3 + \frac{1}{2}}$	
(BDO, SDO, BLO, SLO)	if	$\frac{k_3 - k_1 + 1}{k_3 + \frac{1}{2}} < \theta.$	

Third, we find the average market quality measure and welfare over all the possible values of  $\theta$  at the end of the first trading period (t = 1) when the dark pool is available, given that the optimal

trading strategy when the dark pool was not available,  $D; \mathcal{E}_i^{ND}$ , as:

$$\mathbb{E}_0\left(y_1^{D;\mathcal{E}_i^{ND}}\right) = \int\limits_0^1 y_1^{D,e(\theta;\mathcal{E}_i^{ND})} d\theta,$$

where  $y_1 \in \left\{S_1^{D;\mathcal{E}_i^{ND}}, V_{EX;1}^{D;\mathcal{E}_i^{ND}}, V_{DP;1}^{D;\mathcal{E}_i^{ND}}, V_{T;1}^{D;\mathcal{E}_i^{ND}}, PI_1^{D;\mathcal{E}_i^{ND}}, W_1^{D;\mathcal{E}_i^{ND}}\right\}$ , and where S stands for the inside spread,  $V_{EX}$  the volume traded in the exchange,  $V_{DP}$  the volume traded in the dark pool,  $V_T$  the total traded volume, PI for price informativeness, and W for welfare, and superscript  $D, e(\theta; \mathcal{E}_i^{ND})$  indicates that the variable corresponds to the associated equilibrium strategy profile at t = 1 when rational traders have access to a dark pool, the probability of execution in the dark pool is  $\theta$  and the other parameters are such that when there is no access to the dark pool the equilibrium strategy profile at t = 1 is  $\mathcal{E}_i^{ND}$ , with i = 1, ..., 4.

We denote by  $\mathbb{E}_0\left(S_1^{\mathcal{E}_i^a}\right)$  the *expected inside spread*, where subscript 1 denotes that the measure is computed at the end of the first trading period, t = 1, for a given equilibrium  $\mathcal{E}_i^a$  and define it as  $\mathbb{E}_0\left(S_1^{\mathcal{E}_i^a}\right) = \sum_{\mathcal{B}_1 \in \mathbb{B}} \Pr_1^{\mathcal{E}_i^a}(\mathcal{B}_1)S(\mathcal{B}_1)$ , where  $\mathcal{B}_1 \in \mathbb{B}$  is a possible state of the book and  $a \in \{D, ND\}$ .

Denote  $\mathbb{E}_0\left(V_{EX,1}^{\mathcal{E}_i^a}\right)$  as the expected traded volume in the exchange (subscript EX) in the first trading period for a given equilibrium  $\mathcal{E}_i^a$  and define it as:  $\mathbb{E}_0\left(V_{EX,1}^{\mathcal{E}_i^a}\right) = \sum_{\mathcal{B}_1 \in \mathbb{B}} \Pr_1^{\mathcal{E}_i^a}(\mathcal{B}_1) I_{EX,1}^{\mathcal{E}_i^a}(\mathcal{B}_1)$ , where  $I_{EX,1}^{\mathcal{E}_i^a}(\mathcal{B}_1) = 1$ , if an order has been executed in the exchange during the first trading period when the best prices at the end of the period are  $\mathcal{B}_1$ , and  $I_{EX,1}^{\mathcal{E}_i^a}(\mathcal{B}_1) = 0$  otherwise. Analogously, the expected traded volume in the dark pool at t = 1 for a given equilibrium  $\mathcal{E}_i^a$  is measured as  $\mathbb{E}_0\left(V_{DP,1}^{\mathcal{E}_i^a}\right) = \sum_{\mathcal{B}_1 \in \mathbb{B}} \Pr_1^{\mathcal{E}_i^a}(\mathcal{B}_1) I_{DP,1}^{\mathcal{E}_i^a}(\mathcal{B}_1)$ , where  $I_{DP,1}^{\mathcal{E}_i^a}(\mathcal{B}_1) = 1$  if an order is executed in the dark pool at t = 1 when the best prices at the end of the period are  $\mathcal{B}_1$ , and  $I_{DP,1}^{\mathcal{E}_i^a}(\mathcal{B}_1) = 0$  otherwise. Therefore, the total expected traded volume is:  $\mathbb{E}_0\left(V_{T,1}^{\mathcal{E}_i^a}\right) = \mathbb{E}_0\left(V_{EX,1}^{\mathcal{E}_i^a}\right) + \mathbb{E}_0\left(V_{DP,1}^{\mathcal{E}_i^a}\right)$ .

Denote  $\mathbb{E}_0\left(PI_1^{\mathcal{E}_i^a}\right)$  as expected price informativeness at the end of the first trading period for a given equilibrium  $\mathcal{E}_i^a$  and define it as:  $\mathbb{E}_0\left(PI_1^{\mathcal{E}_i^a}\right) = var\left(v\right) - \sum_{\mathcal{B}_1 \in \mathbb{B}} \Pr_1^{\mathcal{E}_i^a}(\mathcal{B}_1)var(v|\mathcal{B}_1, \mathcal{E}_i^a)$ , where  $var\left(v\right) = \tau^2 k_3^2 = \sigma^2$ , and  $var(v|\mathcal{B}_1, \mathcal{E}_i^a) = \mathbb{E}(v^2|\mathcal{B}_1, \mathcal{E}_i^a) - (\mathbb{E}(v|\mathcal{B}_1, \mathcal{E}_i^a))^2$  is the conditional variance of the value of the asset given the state of the book at the end of the period,  $\mathcal{B}_1$ . Notice that the state of the book at the end of the first period will be different for each equilibrium considered.

Denote  $\mathbb{E}_0\left(W_{1,type}^{\mathcal{E}_i^a}\right)$  as *expected welfare*, where subscript 1 denotes that it is at the end of the first trading period, where subscript *type* denotes whether the trader is informed (*I*), uninformed (*U*), or liquidity trader, and where the superscript indicates the corresponding equilibrium  $\mathcal{E}_i^a$ . We define expected welfare as ex-ante expected profits and, hence,  $\mathbb{E}_0\left(W_{1,type}^{\mathcal{E}_i^a}\right) = \mathbb{E}\left(\Pi_{1,type}^{\mathcal{E}_i^a}\right)$ .

The next proposition finds the effects of adding a dark pool alongside an exchange on market quality and welfare. **Proposition 3** In relation to when there is no access to the dark pool, adding a dark pool alongside an exchange causes the following effects on the market quality parameters at t = 1:

- Expected inside spread:
  - Decreases if the initial equilibrium strategies are (BMO, SMO, BLO, SLO) or (BMO, SMO, NT, NT).
  - Increases if the initial strategy is (BLO, SLO, BLO, BLO).
  - For (BLO, SLO, NT, NT), the change in the expected inside spread depends on the value of π. If π is sufficiently large, then the inside spread increases due to the presence of the dark pool; if π is sufficiently small, then the opposite occurs.
- Expected trading volume:
  - Decreases the expected total trading volume and also decreases the expected trading volume in the exchange if (BMO, SMO, BLO, SLO) or (BMO, SMO, NT, NT).
  - Increases the expected total trading volume but there is no change in the expected trading volume in the exchange if (BLO, SLO, NT, NT) and (BLO, SLO, BLO, SLO)
- Expected price informativeness is always lower due to the existence of the dark pool.
- Expected welfare of rational traders is always higher due to the existence of the dark pool, while the expected welfare for liquidity traders is the same with or without the existence of the dark pool.

Proposition 3 shows at t = 1 some market quality parameters depend on trader or stock market characteristics (i.e., expected inside spread and expected trading volume), while others are unambiguously determined (i.e., expected price informativeness and expected welfare).

Due to the existence of the dark pool expected welfare of each type of market participant at t = 1 is not lower than when the dark pool is unavailable. The ex-ante expected profits of rational traders (informed or uninformed) are always higher with the dark pool compared to when the dark pool is unavailable. Informed traders choose DO when the price improvement outweights the execution risk in the dark and, hence, equilibrium profits of informed traders are higher with dark pool access. In addition, uninformed traders also benefit from the existence of the dark pool since there is a reduction of adverse selection in the exchange when  $\theta$  is high since: (1) if uninformed traders choose LO when there is no access to the dark then uninformed traders continue to submit LO when the dark becomes available. However, when  $\theta$  is high, the profits of uninformed traders are higher due to the reduction of adverse selection in the exchange; (2) if uninformed traders choose NT when there is no access to the dark then uninformed traders switch to LO when the dark becomes available and  $\theta$  is sufficiently large. Hence, the average profits of uninformed traders will be higher when

there is access to the dark pool. Liquidity traders obtain the same expected profits at t = 1 with or without the existence of the dark pool since they only trade according to liquidity needs and do not optimally choose the trading strategy. However, at t = 2 the expected profits of liquidity traders will be affected by the presence of the dark pool, since the profit depends directly on the spread at the end of period t = 1.

Due to the existence of the dark pool, expected price informativeness at t = 1 is always lower compared to when traders do not have access to the dark pool since informed traders migrate to the dark when  $\theta$  is sufficiently large. Therefore, on average, the LOB contains less information with the dark pool than without it.

Adding a dark pool alongside an exchange has an ambiguous effect on the *expected inside spread* at t = 1, which depends on both trader and stock market characteristics. By Corollary 1, we find that if the asset's volatility is high and liquidity is low, or if the probability that a noise trader arrives is low, or if the discount factor is low, then the expected inside spread decreases. This is because, when the dark pool is available, informed traders switch from MO to DO if  $\theta$  is sufficiently high and, hence, expected inside spread decreases since the DO does not change the state of the LOB. Uninformed traders either do not switch their optimal trading strategy (if a LO is submitted) or switch from NT to LO, which decreases the inside spread. In contrast, the expected inside spread increases when the asset's volatility is low and liquidity is high, or if the probability that a noise trader arrives is high, or if the discount factor is high, and adverse selection is low. In contrast, if adverse selection is higher, and the market conditions are such that the optimal trading strategy at t = 1 is (BLO, SLO, NT, NT), then the change in the expected inside spread depends on the degree of adverse selection. If adverse selection is sufficiently high then the inside spread increases, while the converse occurs if adverse selection is sufficiently low. This is because, when  $\theta$  is sufficiently large, the informed switch from LO to DO which increases the inside spread; but the uninformed switch from NT to LO, which decreases the inside spread. Depending on the value of  $\pi$ , one effect might dominate the other.

The effect of adding a dark pool in terms of *expected trading volume* at t = 1 depends on the trader and stock market characteristics which influence the choice of the informed trader in the first round of trading. Specifically, if the asset's volatility is high and liquidity is low, or if the probability that a noise trader arrives is low, or if the discount factor is low then the total expected trading volume and also expected trading volume in the exchange decrease. This is because informed traders' orders migrate to the dark pool if  $\theta$  is sufficiently large, which reduces the expected traded volume in the exchange but also the total expected trading volume since orders that are submitted to the dark pool do not execute at t = 1 with probability  $1 - \theta$ . However, if the asset's volatility is low and liquidity is high, or if the probability that a noise trader arrives is high, or if the discount factor is high then the total expected trading volume increases, but expected trading volume in the exchange remains the same. This occurs since the informed trader migrates to the dark pool when  $\theta$  is sufficiently high, thus augmenting the total expected trading volume in relation to when the

informed submits a LO. The potential switch of the uninformed trader's optimal strategy from NT to LO does not affect the expected volume traded in the exchange at t = 1, and hence, the expected total trading volume remains the same.

## 6 Empirical Implications

In this section we discuss the empirical implications of our model and whether the findings of the empirical literature are consistent with our predictions. These predictions are relevant for the current policy and regulatory debate regarding the effects of dark trading on price discovery, market liquidity and fragmentation of the order flow. The comparative statics that we perform in Section 5 give us two types of predictions: some are true unconditional on the stock market characteristics, while others are true conditional on stock market characteristics. Few of these new cross-sectional empirical predictions are extremely important because they were not previously explored in the previous theoretical models of dark pools, since those models were not delving into the effect of information on the decision to supply or demand liquidity in LOB or migrate to the dark pools.

Our model allows us to derive empirical predictions on the effects of adding a dark pool to a lit exchange based on the comparative statics at t = 1. First of all, there is an empirical prediction that is at the heart of the regulatory debate about whether dark pools increase or reduce price discovery.

#### **Prediction 1.** Adding a dark pool alongside an exchange decreases the informativeness of prices.

Note that our prediction is similar to the theoretical prediction of Ye (2012) and contrary of the one of Zhu (2014) who consider adding a dark pool to a dealer market. The differences between the Ye (2012) and Zhu (2014) are driven by the availability of both informed and uninformed (liquidity) traders to place orders in both the lit and dark venues. Our model features both informed and uninformed traders who can place orders in the two venues but with a different market structure. The possibility to place both market and limit orders changes the order of preferences for the rational traders depending on adverse selection in the LOB (which is a result of the presence of asymmetric information). In addition, since both market and limit order reveal information (Kaniel and Liu, 2006), dark trading may induce a negative effect on price discovery. Our results are in line with the empirical results of Hendershott and Jones (2005), Hatheway et al. (2017), Weaver (2014) and Comerton-Forde and Putniņš (2015) (this last paper only for stocks in which the dark pool trading is large).

Second, our model predicts that adding a dark pool alongside an exchange may increase or decrease market liquidity (measured in our model as the inside spread). Thus, as explained in Proposition 3, it depends whether the trader who migrates to the dark was a supplier or a consumer of liquidity in the exchange. Thus trading in the dark pool can have a positive impact on the spread (reducing it) when the trader who migrates to dark was a consumer of liquidity or a negative impact on spread when he was a supplier of liquidity in the stock exchange. Note that our model without

dark pool implies that the willingness of a trader to demand or supply liquidity depends on stock market-characteristics such as liquidity, volatility, adverse selection or tick size.

# Prediction 2. Cross-section variation with respect to market liquidity and market volatility.

Ceteris paribus, adding a dark pool to a LOB

- where a low liquidity/low volatility stock is traded has a negative impact on inside spread.
- where a midlle liquidity stock/ middle volatility stock is traded has a positive or a negative impact on inside spread depending on the level of adverse selection in that market.
- where a high liquidity stock/high volatility stock is traded has a positive impact on inside spread.

Our theoretical results potentially reconcile the mixed empirical results previously found in the literature. Thus, there are several studies that show that high levels of dark pool trading decreases market liquidity (Degryse et al., 2014; Hatheway et al., 2014; Kwan et al., 2014; Nimalendran and Ray, 2014; Weaver, 2014) while other studies show that dark pool trading increases market liquidity (Buti et al., 2011; Gresse, 2006; Aquilina et al., 2017). Finally, Foley and Putniņš (2016) show that mid-point dark trading in Canadian market does not affect market liquidity.

Note that these papers are developped using very different datasets and this implies that the research questions, the type of data and the regulatory environments are very different. As a result, most of the empirical papers suggest that these differences are determined by the market structure and the financial market regulation that governs a particular market. Interestingly, our model predicts that adding a dark pool can have both a negative and a positive effect on the market performance of the LOB even if the market structure and the regulatory environment are exactly the same. Our results show therefore that cross-sectional characteristics such as market liquidity, volatility or adverse selection are a possible explanation of this heterogeneity in results.

#### Prediction 3. Cross-section variation with respect to tick size.

Ceteris paribus, adding a dark pool to a LOB

- where a low tick size stock is traded has a positive impact on inside spread.
- where a middle tick size stock is traded has a positive or negative impact on inside spread depending on the level of adverse selection in that market.
- where a high tick size stock is traded has a negative impact on inside spread.

Notice that when the tick size is high, the market liquidity is low when there is no dark pool. So, adding a dark pool increases the expected inside spread, as the traders who before provided liquidity by submiting limit orders now prefer to migrate to the dark. This result is similar to the one of Buti et al. (2015) who show that allowing dark orders to "queue-jump" displayed orders reduces traders' willingness to display limit orders on competing lit markets. These results are consistent with Buti et al. (2011) and Kwan et al. (2015) who show that when spreads on traditional exchanges are constrained by minimum pricing increments, traders have incentives to migrate toward dark trading venues since the execution risk in the dark is lower than the execution risk of the limit orders in the exchange. Our results are similar, but the mechanism is different from the one explained by Buti et al. (2011) and Kwan et al. (2015). In our case the tick size does not affect the execution probability but the profits obtained in case of execution. Placing an improving limit order in the case the tick size is large reduces the profits of the limit order and therefore, the traders migh thave more incentives to go to the dark pool.

## Prediction 4. Cross-section variation with respect to the noise and adverse selection in the lit market.

#### Ceteris paribus, adding a dark pool to a LOB

- when the probability of a liquidity trader to arrive is low has a negative impact on inside spread.
- when the probability of a liquidity trader to arrive is high has a positive impact on inside spread when the adverse selection is low and either a positive or a negative impact when the adverse selection is high.

Aquilina et al. (2017) study the other direction of causality: how dark trading affects adverse selection in the lit market. Future empirical studies could analyse using a natural experiment how different levels of adverse selection in the market (measured for example by the probability of informed trading PIN) affect the spread after the introduction of a dark pool.

#### Prediction 5. Cross-section variation with respect to immediacy.

- Adding a dark pool to a LOB when investors are characterized by high immediacy (low δ) has a negative impact on inside spread (all else equal).
- Adding a dark pool to a LOB when investors are characterized by low immediacy (high δ) has a positive impact on inside spread when the adverse selection is low and either positive or negative impact when the adverse selection is high.

## 7 Conclusions

In this paper we study market participants' simultaneous strategic choice of trading venue and order type when traders have access to a dark pool and to an exchange (lit market) that is organized as a limit order book (LOB). We model the exchange as a fully transparent LOB, while the dark pool is an opaque market where orders are executed at the midpoint between the best bid and ask prices prevailing in the exchange. We build a multi-period model that allows us to understand the interaction of the LOB with the dark pool in the presence of asymmetric information. We characterize the Perfect Bayesian Equilibria of the trading game.

We find that adding a dark pool alongside an exchange may shift the optimal strategies of each type of rational trader. In the first period, an uninformed trader may switch from *no trade* to submitting a limit order in the exchange due to the reduction of adverse selection, while the informed trader's strategy diverts from the exchange to the dark pool when the execution risk in the dark pool is sufficiently low. However, uninformed traders may trade in the dark once they have learnt that the value of the asset is favorable from observing the state of the *LOB*. In addition, we find that due to adverse selection, uninformed traders prefer not to trade with informed traders. The optimal strategy of an informed trader in the first period reveals information to uninformed traders about the value of the asset. Our findings show that, even if execution risk in the dark pool is high, informed traders tend to replace market orders by dark orders when they can take advantage of the price improvement.

We also show that adding a dark pool alongside exchange increases the expected welfare of the rational market participants and reduces the expected price informativeness at t = 1. The effect on market liquidity and expected trading volume depend crucially on the stock market characteristics (i.e., liquidity, volatility, adverse selection) and trader characteristics (i.e., discount factor) as these determine if a trader who migrates to the dark pool is a supplier, consumer or does not provide liquidity in the exchange when the dark pool is unavailable. Thus our results help reconcile the positive and negative effects of dark pools on market quality previously found in the empirical studies. Our model provides new testable predictions for cross-sectional analysis. In addition, we provide the asset and trader conditions under which we should expect to observe segmentation of the informed-uninformed order flow. The findings of our paper call for the development of further empirical and experimental work which study the role of information in the competition between a dark pool and an exchange for liquidity.

## References

- [1] Angel, J. 1994. Limit versus market orders. Working Paper, Georgetown University.
- [2] Aquilina, M., Diaz-Rainey, Ibikunle, I. G and Y. Sun. 2017. Dark trading: Transparency and adverse selection in aggregate markets, Working Paper.
- [3] Anand, A., Chakravarty, S., and Martell, T. 2005. Empirical evidence on the evolution of liquidity: Choice of market versus limit orders by informed and uninformed traders. *Journal* of Financial Markets 8, 288–308.
- [4] Barclay, M. J., Hendershott, T., and McCormick, D. T. 2003. Competition among trading venues: Information and trading on electronic communications networks. *Journal of Finance* 58, 2637–2666.
- [5] Banks, E., 2014. Dark pools: Off-Exchange Liquidity in an Era of High Frequency, Program, and Algorithmic Trading. Palgrave Macmillan.
- Biais, B. 1993. Price formation and equilibrium liquidity in fragmented and centralized markets. Journal of Finance 48, 157–185.
- [7] Biais, B., Hillion, P., and Spatt, C. 1995. An empirical analysis of the limit order book and the order flow in the Paris bourse. *Journal of Finance* 50, 1655–1689.
- [8] Biais, B., Martimort, D., and Rochet, J-C. 2000. Competing mechanisms in a common value environment. *Econometrica* 68, 799–837.
- [9] Bloomfield, R., O'Hara, M., and Saar, G. 2005. The "make or take" decision in an electronic market: Evidence on the evolution of liquidity. *Journal of Financial Economics* 75, 165–199.
- [10] Buti, S., Rindi, B., and Werner, I. M. 2011. Diving into dark pools. Fisher College of Business Working Paper.
- [11] Buti, S., F. Consonni, B. Rindi, Y. Wen, and Werner, I. M. 2015. Tick size: theory and evidence. Working paper. University of Toronto.
- [12] Buti, S., Rindi, B., and Werner, I. M. 2017. Dark pool trading strategies, market quality and welfare. *Journal of Financial Economics* 124, 244 – 265.
- [13] Chakravarty, S., and Holden, C. 1995. An integrated model of market and limit orders. Journal of Financial Intermediation 4, 213–241.
- [14] Chowdry, B., and Nanda, V. 1991. Multimarket trading and market liquidity. *Review of Financial Studies* 4, 483–511.

- [15] Comerton-Forde, C., and Putniņš, T. 2015. Dark trading and price discovery. Journal of Financial Economics 118, 70-92.
- [16] Degryse, H., Van Achter, M., and Wuyts, G. 2009. Dynamic order submission strategies with competition between a dealer market and a crossing network. *Journal of Financial Economics* 91(3), 319-338.
- [17] Degryse, H., F. de Jong, and V. v. Kervel. 2014. The impact of dark trading and visible fragmentation on market quality, Review of Finance 19(4), 1587-1622.
- [18] Easley, D., and O'Hara, M. 1991. Order form and information in securities markets. *Journal of Finance* 46, 905–927.
- [19] Foley, S. and Putniņš T. 2016. Should we be afraid of the dark? Dark Trading and market quality. *Journal of Financial Economics* 122, 456-481.
- [20] Foucault, T. 1999. Order flow composition and trading costs in a dynamic limit order market. Journal of Financial Markets 2, 99–134.
- [21] Foucault, T., Kadan, O., and Kandel, E. 2005. Limit order book as a market for liquidity. *Review of Financial Studies* 18, 1171–1217.
- [22] Frutos, M. A., and Manzano, C. 2002. Risk aversion, transparency, and market performance. *Journal of Finance* 57, 959–984.
- [23] Frutos, M. A., and Manzano, C. 2005. Trade disclosure and price experimentation. Journal of Financial Markets 8, 183–216.
- [24] Gomber, P., Sagade, S., Theissen, E., Weber, M.C., and Westheide C. 2016. Competition between equity markets: A review of the consolidation versus fragmentation debate. *Journal* of *Economic Surveys* September, 1-23.
- [25] Glosten, L. R. 1994. Is the electronic open limit order book inevitable? Journal of Finance 49, 1127–1161.
- [26] Glosten, L.R. and Milgrom, P.R. 1985. Bid, ask and transaction prices in a specialist market with heterogeneously informed traders. *Journal of Financial Economics* 14(1), 71-100.
- [27] Goettler, R. L., Parlour, C. A., and Rajan, U. 2009. Informed traders and limit order markets. *Journal of Financial Economics* 93, 67–87.
- [28] Gresse, C. 2006. The effect of crossing-network trading on dealer market's bid-ask spreads, European Financial Management 12, 143-160.
- [29] Harris, L. 1998. Optimal dynamic order submission strategies in some stylized trading problems. Financial Markets, Institutions and Instruments 7, 1–76.

- [30] Hatheway, F., Kwan, A., and Zhen, H. 2017. An empirical analysis of market segmentation on U.S. equities markets. Journal of Financial and Quantitative Analysis (Forthcoming).
- [31] Hendershott, T., and Mendelson, H. 2000. Crossing networks and dealer markets: Competition and performance. *Journal of Finance* 55, 2071–2115.
- [32] Kaniel, R., and Liu, H. 2006. So what orders do informed traders use? Journal of Business 79, 1867–1914.
- [33] Kavajecz, K., and Odders-White, E. 2004. Technical analysis and liquidity provision. *Review of Financial Studies* 17, 1043–1071.
- [34] Kyle, A. S. 1985. Continuous auctions and insider trading. *Econometrica* 53, 1315-1335.
- [35] Kwan, A., Masulis, R., McInish, T.H. 2015. Trading rules, competition for order flow and market fragmentation. *Journal of Financial Economics* 115, 330–348.
- [36] Madhavan, A. 1995. Consolidation, fragmentation, and the disclosure of trading information. *Review of Financial Studies* 8, 579–603.
- [37] Nimalendran, M., Ray, S. 2014. Informational linkages between dark and lit trading venues. Journal of Financial Markets 17, 230–261.
- [38] Pagano, M. 1989. Trading volume and asset liquidity. Quarterly Journal of Economics 104, 255–274.
- [39] Parlour, C. A. 1998. Price dynamics in limit order markets. Review of Financial Studies 11, 789–816.
- [40] Parlour, C. A., and Seppi, D. J. 2003. Liquidity-based competition for order flow. *Review of Financial Studies* 16, 301–343.
- [41] Rock, K. 1996. The specialist's order book and price anomalies. *Review of Financial Studies* 9, 1–20.
- [42] Rosu, I. 2009. A dynamic model of the limit order book. Review of Financial Studies 22, 4601–4641.
- [43] Rosu, I. 2012. Order choice and information in limit order markets. In: Abergel, F., Bouchaud, J., Foucault, T., Lehalle, C.A.,Rosenbaum, M. (eds.), *Market Microstructure: Confronting Many Viewpoints*, Oxford, UK, Wiley 41–60.
- [44] Seppi, D. J. 1997. Liquidity provision with limit orders and a strategic specialist, *Review of Financial Studies* 10, 103–150.

- [45] U.S. Securities and Exchange Commission Public Statement, 2015, Shedding light on dark pools. https://www.sec.gov/news/statement/shedding-light-on-dark-pools.html
- [46] Van Achter, M. 2008. A dynamic limit order market with diversity in trading horizons. Working Paper.
- [47] Weaver, D. G. 2014. The trade-at rule, internalization, and market quality. Working Paper.
- [48] Ye, M. 2011. A glimpse into the dark: Price formation, transaction cost and market share of the crossing network. Working Paper.
- [49] Zhu, H. 2014. Do dark pools harm price discovery? Review of Financial Studies 27, 747-789.

## 8 Appendices

#### A Model without dark pool

**Proof of Proposition 1.** The procedure we follow to check if a particular strategy profile constitutes a PBE is as follows:

- 1. Specify a strategy profile for rational traders at t = 1.
- 2. Update the beliefs of the uninformed trader at t = 2 using Bayes' rule at all information sets, whenever possible.
- 3. Given their beliefs, find the optimal response for the traders at t = 2.
- 4. Given the optimal response of traders at t = 2, find the optimal action for rational traders at t = 1.
- 5. Check if the optimal strategy profile for the traders at t = 1 coincide with the profile suggested in step 1.

We apply the procedure outlined above to check when each possible strategy profile can be an equilibrium.

 $\mathcal{E}_1^{ND}$ : (BMO, SMO, BLO, SLO)

First step. In this case  $\Omega_0 = 0$ ,  $\Omega_1 = 1$ ,  $\Omega_2 = 0$ ,  $\Gamma_0 = 0$ ,  $\Gamma_1 = 0$ , and  $\Gamma_2 = 1$ . Second step. Using Bayes' rule we obtain that  $X = \frac{\lambda \pi}{\eta + \lambda \pi}$  and Y = 0.

Third step. Applying Lemma 1, we know that at t = 2 the optimal strategy of informed traders is to choose a MO, while the optimal strategy of the uninformed trader is as follows:

State of the book	UB	US
$(A_1^1, B_1^1)$	NT	NT
$(A_1^2, B_1^1)$	$\begin{cases} MO & \text{if } \frac{\lambda\pi}{\eta + \lambda\pi} > \frac{k_2}{k_3} \\ NT & \text{if } \frac{\lambda\pi}{\eta + \lambda\pi} \le \frac{k_2}{k_3} \end{cases}$	NT
$(A_1^1, B_1^1 + \tau)$	ŃT	NT
$(A_1^1, B_1^2)$	NT	$\begin{cases} MO & \text{if } \frac{\lambda \pi}{\eta + \lambda \pi} > \frac{k_2}{k_3} \\ NT & \text{if } \frac{\lambda \pi}{\eta + \lambda \pi} \le \frac{k_2}{k_3} \end{cases}$
$(A_1^1 - \tau, B_1^1)$	NT	NT

Table A.1: Optimal responses of uninformed traders at t = 2 when the strategy profile at t = 1 is (BMO, SMO, BLO, SLO).

Fourth step. Given the optimal response of traders at t = 2, we find the optimal action for all rational traders at t = 1.

Informed traders. If they choose a MO, their expected profits equal  $(k_3 - k_1) \tau$ . If, instead, they deviate towards a LO, as they anticipate that uninformed traders choose NT, their expected profits of a LO are  $\delta \frac{\eta}{2} (k_3 + k_1 - 1) \tau$ . Hence, informed traders at t = 1 have no incentives to deviate from the prescribed strategy profile whenever

$$(k_3 - k_1) \tau \ge \delta \frac{\eta}{2} (k_3 + k_1 - 1) \tau$$

Uninformed traders. If they behave as the prescribed profile (LO), then they obtain

$$\frac{\delta}{2} \left( \eta (k_1 - 1) - \lambda \pi \left( k_3 - (k_1 - 1) \right) \right) \tau,$$

given that they anticipate that uninformed traders at t = 2 will choose NT. If, instead, they deviate choosing NT, then they obtain zero profits. Therefore, uninformed traders at t = 1 have no incentives incentive to deviate from the prescribed strategy if and only if

$$\eta(k_1 - 1) - \lambda \pi \left( k_3 - (k_1 - 1) \right) > 0.$$
(5)

**Fifth step.** From the previous two inequalities, nobody at t = 1 has unilateral incentives to deviate from (BMO, SMO, BLO, SLO) whenever

$$k_3 - k_1 \ge \delta \frac{\eta}{2} (k_3 + k_1 - 1)$$
 and  
 $(\lambda \pi + \eta) (k_1 - 1) - \lambda \pi k_3 > 0.$ 

The previous inequality implies that the optimal strategy of an uninformed trader at t = 2 is to choose NT for all possible states of the book.

$$\mathcal{E}_2^{ND}$$
: (BMO, SMO, NT, NT)

First step. In this case  $\Omega_0 = 0$ ,  $\Omega_1 = 1$ ,  $\Omega_2 = 0$ ,  $\Gamma_0 = 1$ ,  $\Gamma_1 = 0$ , and  $\Gamma_2 = 0$ . Second step. Using Bayes's rule we obtain that  $X = \frac{\lambda \pi}{\eta + \lambda \pi}$  and Y is undetermined  $Y \in [0, 1]$  (as Bayes's rule implies  $Y = \frac{0}{0}$ ).

Third step. Applying Lemma 1, we know that at t = 2 the optimal strategy of informed traders is to choose a MO, while the optimal strategy of uninformed trader is as follows:

State of the book	UB	US
$(A_1^1, B_1^1)$	NT	NT
$(A_1^2, B_1^1)$	$\begin{cases} MO & \text{if } \frac{\lambda \pi}{\eta + \lambda \pi} > \frac{k_2}{k_3} \\ NT & \text{if } \frac{\lambda \pi}{\eta + \lambda \pi} \le \frac{k_2}{k_3} \end{cases}$	NT
$(A_1^1, B_1^1 + \tau)$	$\begin{cases} MO & \text{if } Y > \frac{k_1}{k_3} \\ NT & \text{if } Y \le \frac{k_1}{k_3} \end{cases}$	NT
$(A_1^1, B_1^2)$	NT	$\begin{cases} MO & \text{if } \frac{\lambda\pi}{\eta + \lambda\pi} > \frac{k_2}{k_3} \\ NT & \text{if } \frac{\lambda\pi}{\eta + \lambda\pi} \le \frac{k_2}{k_3} \end{cases}$
$(A_1^1 - \tau, B_1^1)$	NT	$\begin{cases} MO & \text{if } Y > \frac{k_1}{k_3} \\ NT & \text{if } Y \le \frac{k_1}{k_3} \end{cases}$

Table A.2: Optimal responses of uninformed traders at t = 2 when the strategy profile at t = 1 is (BMO, SMO, NT, NT).

Fourth step. Given the optimal response of traders at t = 2, we find the optimal action of rational traders at t = 1.

Informed traders. If they choose a MO, they obtain  $(k_3 - k_1)\tau$ . If, instead, they deviate towards a LO, as they anticipate that uninformed traders will not choose a MO of different sign at t = 2, then their expected profits will be:  $\delta \frac{\eta}{2} (k_3 + k_1 - 1)\tau$ . Hence, informed traders at t = 1 have no incentives to deviate from the prescribed strategy profile whenever

$$k_3 - k_1 \ge \delta \frac{\eta}{2} (k_3 + k_1 - 1).$$

Uninformed traders. If they behave as the prescribed profile (NT), then they obtain 0. If, instead, they deviate choosing a LO, as they anticipate that uninformed traders will not choose a MO of different sign at t = 2, their expected profits will be

$$\frac{\delta}{2} \left( \eta (k_1 - 1) - \lambda \pi \left( k_3 - (k_1 - 1) \right) \right) \tau.$$

Therefore, uninformed traders at t = 1 have no incentives to deviate from the prescribed strategy if and only if

$$0 \ge (\lambda \pi + \eta) \left(k_1 - 1\right) - \lambda \pi k_3. \tag{6}$$

**Fifth step.** No trader at t = 1 has unilateral incentives to deviate from (BMO, SMO, NT, NT) if and only if

$$k_3 - k_1 \ge \delta \frac{\eta}{2} (k_3 + k_1 - 1)$$
 and  
 $0 \ge (\lambda \pi + \eta) (k_1 - 1) - \lambda \pi k_3.$ 

 $\mathcal{E}_3^{ND}$ : (BLO, SLO, BLO, BLO)

First step. In this case  $\Omega_0 = 0$ ,  $\Omega_1 = 0$ ,  $\Omega_2 = 1$ ,  $\Gamma_0 = 0$ ,  $\Gamma_1 = 0$ , and  $\Gamma_2 = 1$ . Second step. Using Bayes's rule we obtain that X = 0 and  $Y = \pi$ .

Third step. Applying Lemma 1, we know that at t = 2 the optimal strategy for informed traders is to choose a MO, while for the uninformed trader is as follows:

State of the book	UB	US
$(A_1^1, B_1^1)$	NT	NT
$(A_1^2, B_1^1)$	NT	NT
$(A_1^1, B_1^1 + \tau)$	$\begin{cases} MO & \text{if } \pi > \frac{k_1}{k_3} \\ NT & \text{if } \pi \le \frac{k_1}{k_3} \end{cases}$	NT
$(A_1^1, B_1^2)$	NT	NT
$(A_1^1 - \tau, B_1^1)$	NT	$\begin{cases} MO & \text{if } \pi > \frac{k_1}{k_3} \\ NT & \text{if } \pi \le \frac{k_1}{k_3} \end{cases}$

Table A.3: Optimal responses of uninformed traders at t = 2 when the strategy profile at t = 1 is (BLO, SLO, BLO, BLO).

Fourth step. Given the optimal response of traders at t = 2, we find the optimal action for the rational traders at t = 1.

Informed traders. If they choose a LO, as they anticipate that uninformed traders will not choose a MO of different sign at t = 2, then their expected profits will be:  $\delta \frac{\eta}{2} (k_3 + k_1 - 1) \tau$ . If, instead, they choose a market order, then they obtain  $(k_3 - k_1) \tau$ . Hence, at t = 1 informed traders have no incentives to deviate from the prescribed strategy profile whenever

$$\delta \frac{\eta}{2} \left( k_3 + k_1 - 1 \right) > k_3 - k_1$$

Uninformed traders. If they select a LO, their expected profits are  $\frac{\delta \tau}{2} (\eta(k_1 - 1) - \lambda \pi (k_3 - (k_1 - 1)))$ since they anticipate that traders will not choose a MO of different sign at t = 2. If, instead, they choose NT, they obtain null profits. Therefore, at t = 1 uninformed traders have no incentives to deviate from the prescribed strategy if and only if

$$\left(\lambda\pi + \eta\right)\left(k_1 - 1\right) - \lambda\pi k_3 > 0. \tag{7}$$

Fifth step. No trader at t = 1 has unilateral incentives to deviate from (BLO, SLO, BLO, SLO) if and only if

$$\delta \frac{\eta}{2} (k_3 + k_1 - 1) > k_3 - k_1$$
 and  
 $(\lambda \pi + \eta) (k_1 - 1) - \lambda \pi k_3 > 0.$ 

 $\mathcal{E}_4^{ND}$ : (BLO, SLO, NT, NT)

First step. In this case  $\Omega_0 = 0$ ,  $\Omega_1 = 0$ ,  $\Omega_2 = 1$ ,  $\Gamma_0 = 1$ ,  $\Gamma_1 = 0$ , and  $\Gamma_2 = 0$ .

**Second step.** Using Bayes's rule we obtain that X = 0 and Y = 1.

Third step. Applying Lemma 1, we know that at t = 2 the optimal strategy for informed traders is to choose a MO, while for the uninformed trader is as follows:

State of the book	UB	$\mathbf{US}$
$(A_1^1, B_1^1)$	NT	NT
$(A_1^2, B_1^1)$	NT	NT
$(A_1^1, B_1^1 + \tau)$	MO	NT
$(A_1^1, B_1^2)$	NT	NT
$(A_1^1 - \tau, B_1^1)$	NT	MO

Table A.4: Optimal responses of uninformed traders at t = 2 when the strategy profile at t = 1 is (BLO, SLO, NT, NT).

Fourth step. Given the optimal response of traders at t = 2, find the optimal action for the rational traders at t = 1.

• Informed traders. As they anticipate that uninformed traders will not choose a MO of different sign at t = 2, then their expected profits are:  $\delta \frac{\eta}{2} (k_3 + k_1 - 1) \tau$ . If, instead, they deviate towards a MO, they obtain  $(k_3 - k_1) \tau$ . Hence, informed traders have no incentives to deviate from the prescribed strategy profile whenever

$$\delta \frac{\eta}{2} \left( k_3 + k_1 - 1 \right) \tau > \left( k_3 - k_1 \right) \tau.$$

• Uninformed traders. If they chooses NT, they obtain zero profits. If, instead, an uninformed buyer (seller) deviate towards a *BLO* (*SLO*), then he obtains  $\frac{\delta \tau}{2} ((\lambda \pi + \eta) (k_1 - 1) - \lambda \pi k_3)$ . Hence, uninformed traders have no incentive to deviate from the prescribed strategy profile whenever

$$0 \ge \frac{\delta}{2} \left( \eta(k_1 - 1) - \lambda \pi \left( k_3 - (k_1 - 1) \right) \right).$$
(8)

**Fifth step.** Nobody at t = 1 has unilateral incentives to deviate from (BLO, SLO, NT, NT) whenever

$$\delta \frac{\eta}{2} (k_3 + k_1 - 1) > (k_3 - k_1)$$
  
$$0 \ge \frac{\delta}{2} (\eta (k_1 - 1) - \lambda \pi (k_3 - (k_1 - 1))).$$

**Lemma 5** When there is no dark pool and  $k_1 = 1$  the uninformed trader at t = 1 always chooses NT.

**Proof.** The previous lemma implies that the strategies (BMO, SMO, BLO, SLO) and (BLO, SLO, BLO, BLO) cannot be equilibria of the game.

**Proposition 4** If  $k_1 = 1$ , then a PBE of the game is as follows:

• (BMO, SMO, NT, NT) is the optimal strategy profile for traders at t = 1 if

$$k_3 - 1 \ge \delta \frac{\eta}{2} k_3.$$

The beliefs of uninformed traders at t = 2 are:  $X = \frac{\lambda \pi}{\eta + \lambda \pi}$  and  $Y = p \in [0, 1]$ . The optimal strategy of informed traders at t = 2 is to choose MO for all possible states of the book and the optimal strategies of uninformed traders at t = 2 are described in Table A.2.

• (BLO, SLO, NT, NT) is the optimal strategy profile for traders at t = 1 if

$$\delta\left(\frac{\eta}{2} + \frac{(1-\pi)\lambda}{2}\right)k_3 > k_3 - 1.$$

The beliefs of uninformed traders at t = 2 are: X = 0 and Y = 1. The optimal strategy of informed traders at t = 2 is to choose MO for all possible states of the book and the optimal strategies of uninformed traders at t = 2 are described in Table A.4.

**Proof of Proposition 4.** Note that when we replace  $k_1 = 1$  in the Proof of Proposition 1 the conditions (5) and (7) are never satisfied and therefore the strategies (*BMO*, *SMO*, *BLO*, *SLO*) and (*BLO*, *SLO*, *BLO*, *BLO*) cannot be part of an equilibrium of the game.

Moreover when  $k_1 = 1$ , the conditions (6) and (8) are always satisfied.

#### **B** Model with dark pool

**Proof of Lemma 3.** By simply inspection of the payoff in Table (6) can be seen that the informed buyers at t = 2 never choose NT because this order is dominated by placing a MO. Notice also that the uninformed traders never select a DO and therefore  $\Gamma_3 = 0$  which implies

$$P_I = p_{BLO,2}^{IH} \left( \mathcal{B}_1 = \varnothing \right) = p_{SLO,2}^{IL} \left( \mathcal{B}_1 = \varnothing \right) = 0.$$

Consequently, the informed traders never choose a LO at t = 2, since this order is also dominated by a MO.

Let us determine next the optimal strategy for each trader. Depending on the values of the parameters we have 6 possible cases for the informed trader and 16 for the uninformed trader.

First let us first focus on the informed traders. Note that since  $k_3 > k_2 > k_1 \ge 1$ , the following inequalities hold

$$\frac{k_3 - k_2}{k_3 - \frac{k_2 - k_1}{2}} < \frac{k_3 - k_1}{k_3 + \frac{k_2 - k_1}{2}} < \frac{k_3 - k_1}{k_3} < \frac{k_3 - k_1}{k_3 - \frac{1}{2}} < \frac{k_3 - k_1 + 1}{k_3 + \frac{1}{2}}.$$

We define by

$$\theta_X \equiv \frac{Xk_3 - k_2}{Xk_3 - \frac{k_2 - k_1}{2}}$$
$$\theta_Y \equiv \frac{Yk_3 - k_1}{Yk_3 - \frac{1}{2}}.$$

The optimal strategies of the informed traders are:

Condition	Optimal Strategies of Informed Traders at t=2		
	State of the Book	IH	IL
$Case I_1$ $\theta \le \frac{k_3 - k_2}{k_3 - \frac{k_2 - k_1}{2}}$	$\begin{array}{c}(A_1^1,B_1^1)\\(A_1^2,B_1^1)\\(A_1^1,B_1^1+\tau)\\(A_1^1,B_1^2)\\(A_1^1-\tau,B_1^1)\end{array}$	BMO BMO BMO BMO BMO	SMO SMO SMO SMO SMO
Case $I_2$ $\frac{k_3 - k_2}{k_3 - \frac{k_2 - k_1}{2}} < \theta$ $\leq \frac{k_3 - k_1}{k_3 + \frac{k_2 - k_1}{2}}$	$\begin{array}{c} (A_1^1,B_1^1) \\ (A_1^2,B_1^1) \\ (A_1^1,B_1^1+\tau) \\ (A_1^1,B_1^2) \\ (A_1^1-\tau,B_1^1) \end{array}$	BMO BDO BMO BMO BMO	SMO SMO SMO SDO SMO
$\begin{array}{l} \text{Case } I_3\\ \frac{k_3-k_1}{k_3+\frac{k_2-k_1}{2}} < \theta\\ \leq \frac{k_3-k_1}{k_3} \end{array}$	$egin{aligned} &(A_1^1,B_1^1)\ &(A_1^2,B_1^1)\ &(A_1^1,B_1^1+ au)\ &(A_1^1,B_1^1+ au)\ &(A_1^1,B_1^2)\ &(A_1^1- au,B_1^1) \end{aligned}$	BMO BDO BMO BDO BMO	SMO SDO SMO SDO SMO
$\begin{array}{l} \text{Case } I_4 \\ \frac{k_3 - k_1}{k_3} < \theta \\ \leq \frac{k_3 - k_1}{k_3 - \frac{1}{2}} \end{array}$	$\begin{array}{c} (A_1^1,B_1^1) \\ (A_1^2,B_1^1) \\ (A_1^1,B_1^1+\tau) \\ (A_1^1,B_1^2) \\ (A_1^1,-\tau,B_1^1) \end{array}$	BDO BDO BMO BDO BMO	SDO SDO SMO SDO SMO
Case $I_5$ $\frac{k_3-k_1}{k_3-\frac{1}{2}} < \theta$ $\leq \frac{k_3-k_1+1}{k_3+\frac{1}{2}}$	$\begin{array}{c}(A_1^1,B_1^1)\\(A_1^2,B_1^1)\\(A_1^1,B_1^1+\tau)\\(A_1^1,B_1^2)\\(A_1^1,-\tau,B_1^1)\end{array}$	BDO BDO BDO BDO BMO	SDO SDO SMO SDO SDO
$\begin{array}{l} \text{Case } I_6 \\ \frac{k_3-k_1+1}{k_3+\frac{1}{2}} < \theta \end{array}$	$egin{array}{c} (A_1^1,B_1^1)\ (A_1^2,B_1^1)\ (A_1^1,B_1^1+ au)\ (A_1^1,B_1^2)\ (A_1^1,B_1^2)\ (A_1^1- au,B_1^1) \end{array}$	BDO BDO BDO BDO BDO	SDO SDO SDO SDO SDO

Table B.1: Optimal Strategies of Informed Traders at  $t=2\,$ 

Condition	Optimal Strateg	ies of Uninformed '	Traders at $t = 2$
	State of the Book	UB	US
	$(A_1^1, B_1^1)$	NT	NT
Case U.1.1	$(A_1^2, B_1^1)$	NT	SDO
$Xk_3 < \frac{k_2 - k_1}{2} < k_2$	$(A_1^1, B_1^1 + \tau)$	NT	SDO
$Yk_3 < \frac{1}{2}(< k_1)$	$(A_1^1, B_1^2)$	BDO	NT
_	$(A_1^1 - \tau, B_1^1)$	BDO	NT
	$(A_1^1, B_1^1)$	NT	NT
Case U.1.2	$(A_1^2, B_1^1)$	NT	SDO
$Xk_3 < \frac{k_2 - k_1}{2} < k_2$	$(A_1^1, B_1^1 + \tau)$	NT	NT
$Yk_3 = \frac{1}{2}(< k_1)$	$(A_1^1, B_1^2)$	BDO	NT
	$(A_1^1 - \tau, B_1^1)$	NT	NT
	$(A_1^1, B_1^1)$	NT	NT
Case U.1.3	$(A_1^2, B_1^1)$	NT	SDO
$Xk_3 < \frac{k_2 - k_1}{2} < k_2$	$(A_1^1, B_1^1 + \tau)$	BDO	NT
$\frac{1}{2} < Yk_3 \le k_1$	$(A_1^1, B_1^2)$	BDO	NT
	$(A_1^1 - \tau, B_1^1)$	NT	SDO
	$(A_1^1, B_1^1)$	NT	NT
Case U.1.4	$(A_1^2, B_1^1)$	NT	SDO
$Xk_3 < \frac{k_2 - k_1}{2} < k_2$	$(A_1^1, B_1^1 + \tau)$	$\begin{cases} BDO  \theta > \theta_Y \\ \end{array}$	NT
		$\bigcup BMO  \theta \le \theta_Y$	
$k_1 < Yk_3$	$(A_1^1, B_1^2)$	BDO	NT
	$(A_1^1 - \tau, B_1^1)$	NT	$\left  \begin{array}{c} SDO  \theta > \theta_Y \\ SDO  \theta > \theta_Y \end{array} \right $
			$\mid \bigcup SMO  \theta \leq \theta_Y$

Table B.2: Optimal Strategies of Uninformed Traders at  $t=2\,$ 

Condition	Optimal Strateg	ies of Uninformed '	Traders at $t = 2$
	State of the Book	UB	US
	$(A_1^1, B_1^1)$	NT	NT
Case U.2.1	$(A_1^2, B_1^1)$	NT	NT
$Xk_3 = \frac{k_2 - k_1}{2} < k_2$	$(A_1^1, B_1^1 + \tau)$	NT	SDO
$Yk_3 < \frac{1}{2} (< k_1)$	$(A_1^1, B_1^2)$	NT	NT
_	$(A_1^1 - \tau, B_1^1)$	BDO	NT
	$(A_1^1, B_1^1)$	NT	NT
Case U.2.2	$(A_1^2, B_1^1)$	NT	NT
$Xk_3 = \frac{k_2 - k_1}{2} < k_2$	$(A_1^1, B_1^1 + \tau)$	NT	NT
$Yk_3 = \frac{1}{2}(< k_1)$	$(A_1^1, B_1^2)$	NT	NT
_	$(A_1^1 - \tau, B_1^1)$	NT	NT
	$(A_1^1, B_1^1)$	NT	NT
Case U.2.3	$(A_1^2, B_1^1)$	NT	NT
$Xk_3 = \frac{k_2 - k_1}{2} < k_2$	$(A_1^1, B_1^1 + \tau)$	BDO	NT
$\frac{1}{2} < Yk_3 \le k_1$	$(A_1^1, B_1^2)$	NT	NT
	$(A_1^1 - \tau, B_1^1)$	NT	SDO
	$(A_1^1, B_1^1)$	NT	NT
Case U.2.4	$(A_1^2, B_1^1)$	NT	NT
$Xk_3 < \frac{k_2 - k_1}{2} < k_2$	$(A_1^1, B_1^1 + \tau)$	$\left\{\begin{array}{ll}BDO  \theta > \theta_Y\\BMO  \theta \le \theta_Y\end{array}\right.$	NT
$k_1 < Yk_3$	$(A_1^1, B_1^2)$	NT	NT
	$(A_1^1 - \tau, B_1^1)$	NT	$\left\{\begin{array}{ll}SDO & \theta > \theta_Y\\SMO & \theta \le \theta_Y\end{array}\right.$

Optimal Strategies of Uninformed Traders at t = 2 (Continuation)

Condition	Optimal Strateg	ies of Uninformed '	Traders at $t = 2$
	State of the Book	UB	US
		NG	NG
	$(A_1^1, B_1^1)$		NT
Case U.3.1	$(A_1^2, B_1^1)$	BDO	NT
$\frac{k_2 - k_1}{2} < Xk_3 \le k_2$	$(A_1^1, B_1^1 + \tau)$	NT	SDO
$Yk_3 < \frac{1}{2}$	$(A_1^1, B_1^2)$	NT	SDO
	$(A_1^1 - \tau, B_1^1)$	BDO	NT
	$(A_1^1, B_1^1)$	NT	NT
Case U.3.2	$(A_1^2, B_1^1)$	BDO	NT
$\frac{k_2 - k_1}{2} < Xk_3 \le k_2$	$(A_1^1, B_1^1 + \tau)$	NT	NT
$Yk_3 = \frac{1}{2}$	$(A_1^1, B_1^2)$	NT	SDO
	$(A_1^1 - \tau, B_1^1)$	NT	NT
	$(A_1^1, B_1^1)$	NT	NT
Case U.3.3	$(A_1^2, B_1^1)$	BDO	NT
$\frac{k_2 - k_1}{2} < X k_3 \le k_2$	$(A_1^1, B_1^1 + \tau)$	BDO	NT
$\frac{1}{2} < Yk_3 \le k_1$	$(A_1^1, B_1^2)$	NT	SDO
	$(A_1^1 - \tau, B_1^1)$	NT	SDO
	$(A_1^1, B_1^1)$	NT	NT
Case U.3.4	$(A_1^2, B_1^1)$	BDO	NT
$\frac{k_2 - k_1}{2} < Xk_3 \le k_2$	$(A_1^1, B_1^1 + \tau)$	$\begin{cases} BDO  \theta > \theta_Y \\ BMO  \theta < \theta_V \end{cases}$	NT
$k_1 < Yk_3$	$(A_1^1, B_1^2)$	NT	SDO
	$(A_1^1 - \tau, B_1^1)$	NT	$\begin{cases} SDO  \theta > \theta_Y \\ SMO  \theta \le \theta_Y \end{cases}$

Optimal Strategies of Uninformed Traders at t = 2 (Continuation)

Condition	Optimal Strategies of Uninformed Traders at $t = 2$		
	State of the Book	UB	US
Case U.4.1 $k_2 < Xk_3$ $Yk_3 < \frac{1}{2}$	$(A_1^1, B_1^1) (A_1^2, B_1^1) (A_1^1, B_1^1 + \tau) (A_1^1, B_1^2) (A_1^1 - \tau, B_1^1)$	$\begin{cases} NT\\ BDO & \text{if } \theta > \theta_X\\ BMO & \text{if } \theta \le \theta_X\\ NT\\ NT\\ BDO \end{cases}$	$NT$ $NT$ $SDO$ $\begin{cases} SDO & \text{if } \theta > \theta_X \\ SMO & \text{if } \theta \le \theta_X \\ NT \end{cases}$
Case U.4.2 $k_2 < Xk_3$ $Yk_3 = \frac{1}{2}$	$(A_1^1, B_1^1) \\ (A_1^2, B_1^1) \\ (A_1^1, B_1^1 + \tau) \\ (A_1^1, B_1^2) \\ (A_1^1 - \tau, B_1^1) $	$\begin{cases} NT \\ BDO & \text{if } \theta > \theta_X \\ BMO & \text{if } \theta \le \theta_X \\ NT \\ NT \\ NT \\ NT \end{cases}$	$NT$ $NT$ $NT$ $SDO  ext{ if } \theta > \theta_X$ $SMO  ext{ if } \theta \le \theta_X$ $NT$
Case U.4.3 $k_2 < Xk_3$ $\frac{1}{2} < Yk_3 \le k_1$	$\begin{array}{c} (A_1^1,B_1^1)\\ (A_1^2,B_1^1)\\ (A_1^1,B_1^1+\tau)\\ (A_1^1,B_1^2)\\ (A_1^1,B_1^2)\\ (A_1^1-\tau,B_1^1)\end{array}$	$NT$ $\begin{cases} BDO & \text{if } \theta > \theta_X \\ BMO & \text{if } \theta \le \theta_X \\ BDO \\ NT \\ NT \end{cases}$	$NT$ $NT$ $NT$ $SDO  \text{if } \theta > \theta_X$ $SMO  \text{if } \theta \le \theta_X$ $SDO$
Case U.4.4 $k_2 < Xk_3$ $k_1 < Yk_3$	$(A_1^1, B_1^1) (A_1^2, B_1^1) (A_1^1, B_1^1 + \tau) (A_1^1, B_1^2) (A_1^1 - \tau, B_1^1)$	$NT$ $\begin{cases} BDO & \text{if } \theta > \theta_X \\ BMO & \text{if } \theta \le \theta_X \\ BDO & \text{if } \theta > \theta_Y \\ BMO & \text{if } \theta \le \theta_Y \\ NT \\ NT \\ NT \end{cases}$	$NT$ $NT$ $NT$ $\begin{cases} SDO & \text{if } \theta > \theta_X \\ SMO & \text{if } \theta \le \theta_X \\ SDO & \text{if } \theta > \theta_Y \\ SMO & \text{if } \theta \le \theta_Y \end{cases}$

Optimal Strategies of Uninformed Traders at t = 2 (Continuation)

**Proof of Proposition 2.** Because of the symmetry of the model, without any loss of generality, at t = 1 we focus on buyers. We present the proof for one of the possible strategy profile at t = 1 that yields an equilibrium. The proofs of all the other 5 equilibria can be obtained on request from the authors. Note that in all equilibria the optimal responses of informed traders at t = 2 are given in Table B.1.

 $\mathcal{E}_1^D$ : (BMO, SMO, BLO, SLO)

First step. In this case  $\Omega_0 = 0, \Omega_1 = 1, \Omega_2 = 0, \Omega_3 = 0, \Gamma_0 = 0, \Gamma_1 = 0, \Gamma_2 = 1$ , and  $\Gamma_3 = 0$ . Second step. Using Bayes's rule,

$$X = \frac{\lambda \pi}{\eta + \lambda \pi}, Y = 0 \text{ and } Z = q \in [0, 1]$$

Third step. Using steps 1 and 2, the expected profits of uninformed traders at t = 2 are given by

UB	BMO	BDO	BLO	NT
$(A_1^1, B_1^1)$	$-k_1 au$	0	0	0
$(A_1^2, B_1^1)$	$\left(\frac{\lambda\pi}{\eta+\lambda\pi}k_3-k_2\right) au$	$ heta\left(rac{\lambda\pi}{\eta+\lambda\pi}k_3-rac{k_2-k_1}{2} ight) au$	0	0
$(A_1^1, B_1^1 + \tau)$	$-k_1\tau$	$-\frac{\theta}{2}\tau$	0	0
$(A_1^1, B_1^2)$	$-\left(rac{\lambda\pi}{\eta+\lambda\pi}k_3+k_1 ight) au$	$- heta\left(rac{\lambda\pi}{\eta+\lambda\pi}k_3-rac{k_2-k_1}{2} ight) au$	0	0
$(A_1^1 - \tau, B_1^1)$	$-(k_1-1)\tau$	$\frac{ heta}{2} au$	0	0

Table B.3: Expected profits for an uninformed buyer at t = 2 when the strategy profile at t = 1 is (BMO, SMO, BLO, SLO).

US	SMO	SDO	SLO	NT
$(A_1^1, B_1^1)$	$-k_1 au$	0	0	0
$(A_1^2, B_1^1)$	$-\left(k_1+\frac{\lambda\pi}{\eta+\lambda\pi}k_3\right) au$	$-\theta\left(\frac{\lambda\pi}{\eta+\lambda\pi}k_3-\frac{k_2-k_1}{2}\right) au$	0	0
$(A_1^1, B_1^1 + \tau)$	$-(k_1-1)\tau$	$\frac{\theta \tau}{2}$	0	0
$(A_1^1, B_1^2)$	$\left(\frac{\lambda\pi}{\eta+\lambda\pi}k_3-k_2\right) au$	$\theta\left(rac{\lambda\pi}{\eta+\lambda\pi}k_3-rac{k_2-k_1}{2} ight) au$	0	0
$(A_1^1 - \tau, B_1^1)$	$-k_1\tau$	$-\frac{\theta \tau}{2}$	0	0

Table B.4: Expected profits for an uninformed seller at t = 2 when the strategy profile at t = 1 is (BMO, SMO, BLO, SLO).

Hence, the optimal responses of uninformed traders are:

State of the book	UB	US
$(A_1^1, B_1^1)$	NT	NT
$(A_1^2, B_1^1)$	$\begin{cases} NT & \text{if } Xk_3 \leq \frac{k_2 - k_1}{2} \\ BDO & \begin{cases} \text{if } \frac{k_2 - k_1}{2} < Xk_3 \leq k_2 \text{ or} \\ \text{if } k_2 < Xk_3 \text{ and } \theta > \theta_X \\ BMO & \text{if } k_2 < Xk_3 \text{ and } \theta \leq \theta_X \end{cases}$	$\begin{cases} SDO \text{ if } Xk_3 < \frac{k_2 - k_1}{2} \\ NT  \text{if } \frac{k_2 - k_1}{2} \le Xk_3 \end{cases}$
$(A_1^1, B_1^1 + \tau)$	NT	SDO
$(A_1^1, B_1^2)$	$\begin{cases} BDO \text{ if } Xk_3 < \frac{k_2 - k_1}{2} \\ NT  \text{if } \frac{k_2 - k_1}{2} \le Xk_3 \end{cases}$	$\begin{cases} NT & \text{if } Xk_3 \leq \frac{k_2 - k_1}{2} \\ SDO & \begin{cases} \text{if } \frac{k_2 - k_1}{2} < Xk_3 \leq k_2 \text{ or} \\ \text{if } k_2 < Xk_3 \text{ and } \theta > \theta_X \\ SMO & \text{if } k_2 < Xk_3 \text{ and } \theta \leq \theta_X \end{cases}$
$(A_1^1 - \tau, B_1^1)$	BDO	NT

Table B.5: Optimal responses of uninformed traders at t = 2 when the strategy profile at t = 1 is (BMO, SMO, BLO, SLO).

Fourth step. Given the optimal responses of rational traders at t = 2, we find the optimal action for the rational traders at t = 1 in each of the 6 cases. However, given the nature of this particular equilibrium, we can group cases into the following and analyze them:

**Case**  $I_1 + I_2 + I_3 : \theta \le \frac{k_3 - k_1}{k_3}$ 

• Informed traders

Consider an informed buyer at t = 1. If he chooses a *BMO*, then he obtains

$$\mathbb{E}\left(\Pi_{BMO,1}^{IH}\right) = \left(k_3 - k_1\right)\tau.$$

If instead he deviates towards a *BLO*, then in the next period the prices will be  $(A_1^1, B_1^1 + \tau)$  and, then, he anticipates the following behavior for potential sellers at t = 2:

- 1. if there is an uninformed seller, then he will choose SDO, and
- 2. if there is a liquidity seller, then he will place a SMO.

Therefore, the *BLO* at t = 1 will only be executed if in the next period there is a liquidity seller. Thus, the corresponding expected profits are given by

$$\mathbb{E}\left(\Pi_{BLO,1}^{IH}\right) = \frac{\eta}{2}\delta\left(k_3 + k_1 - 1\right)\tau.$$

If instead he deviates towards a BDO, he knows that in the next period the prices in the book will not change. In this case, he anticipates the following behavior for traders at t = 2:

- 1. if there is an informed trader, then he will be a buyer and will choose a BMO,
- 2. if there is an uninformed buyer, then he will choose a NT, and
- 3. if there is an uninformed seller, then he will choose a NT.

Thus, Table 8 implies that

$$\mathbb{E}\left(\Pi_{BDO,1}^{IH}\right) = \theta k_3 \tau + (1-\theta)\delta^2 \left(k_3 - k_1 - (k_2 - k_1)\left(\lambda \pi + \frac{\eta}{2}\right)\right)\tau,$$

since  $I_{SLO,2}^{US,\mathcal{B}_1=\emptyset} = 0$  and  $I_{BMO,2}^{IH,\mathcal{B}_1=\emptyset} = 1$ . Hence, informed traders at t = 1 have no incentives to deviate from the prescribed strategy profile whenever

$$k_3 - k_1 \ge \frac{\eta}{2} \delta \left( k_3 + k_1 - 1 \right) \text{ and} k_3 - k_1 \ge \theta k_3 + (1 - \theta) \delta^2 \left( (k_3 - k_1) - (k_2 - k_1) \left( \lambda \pi + \frac{\eta}{2} \right) \right)$$

• Uninformed traders

Consider an uninformed buyer at t = 1. If instead he deviates towards a *BLO*, then in the next period the prices will be  $(A_1^1, B_1^1 + \tau)$  and, then, he anticipates the following behavior for potential sellers at t = 2:

- 1. if there is an informed seller, then he chooses SMO,
- 2. if there is an uninformed seller, then he will choose SDO, and
- 3. if there is a liquidity seller, then he will set SMO.

Therefore, the BLO will be executed only with a liquidity seller or with a informed seller. Hence, the corresponding expected profits will be

$$\mathbb{E}\left(\Pi_{BLO,1}^{UB}\right) = \frac{\delta}{2}\left(\left(\lambda\pi + \eta\right)\left(k_1 - 1\right) - \lambda\pi k_3\right)\tau,$$

since  $I_{SMO,2}^{IL,\mathcal{B}_1=BLO} = 1.$ 

If instead, the uninformed buyer deviates and chooses BDO, then he knows that at t = 2 the prices in the book will not change. In this case, he anticipates the following behavior for traders at t = 2:

1. if there is an informed buyer, then he will choose a BMO,

- 2. if there is an informed seller, then he will choose a SMO,
- 3. if there is an uninformed buyer, then he will choose NT, and
- 4. if there is an uninformed seller, then he will choose NT.

Thus, Table 9 implies that

$$\mathbb{E}\left(\Pi_{BDO,1}^{UB}\right) = (1-\theta)\delta^2\left(\frac{\lambda}{2}\pi \left(k_1 - k_2\right) + \frac{\eta}{2}\left(k_1 - k_2\right) - k_1\right)\tau < 0,$$

since  $I_{SLO,2}^{IL,\mathcal{B}_1=\varnothing} = I_{SLO,2}^{US,\mathcal{B}_1=\varnothing} = 0$  and  $I_{BMO,2}^{IH,\mathcal{B}_1=\varnothing} = 1$ .

If instead, the uninformed buyer deviates and chooses NT, then he will obtain

$$\mathbb{E}\left(\Pi_{NT,1}^{UB}\right) = 0.$$

Hence, he will not have incentives to deviate if

$$\left(\lambda \pi + \eta\right) \left(k_1 - 1\right) - \lambda \pi k_3 > 0.$$

Fifth step. From step 4, nobody at t = 1 has unilateral incentives to deviate whenever

$$k_{3} - k_{1} \ge \frac{\eta}{2} \delta \left( k_{3} + k_{1} - 1 \right),$$
  

$$k_{3} - k_{1} \ge \theta k_{3} + (1 - \theta) \delta^{2} \left( k_{3} - k_{1} - (k_{2} - k_{1}) \left( \lambda \pi + \frac{\eta}{2} \right) \right), \text{ and}$$
  

$$(\lambda \pi + \eta) \left( k_{1} - 1 \right) - \lambda \pi k_{3} > 0.$$

**Case**  $I_4 + I_4 + I_6 : \frac{k_3 - k_1}{k_3} < \theta$ 

• Informed traders

Consider an informed buyer at t = 1. If he chooses a *BMO*, then he obtains

$$\mathbb{E}\left(\Pi_{BMO,1}^{IH}\right) = \left(k_3 - k_1\right)\tau_1$$

If instead he deviates towards a BDO, he knows that in the next period the prices in the book will not change. In this case, he anticipates the following behavior for rational traders at t = 2:

- 1. if there is an informed trader, then it will be a buyer and will choose a BDO,
- 2. if there is an uninformed buyer, then he will choose a NT, and
- 3. if there is an uninformed seller, then he will choose a NT.

Therefore, it follows that

$$\mathbb{E}\left(\Pi_{BDO,1}^{IH}\right) = \theta k_3 \tau + (1-\theta)\delta^2 \left(k_3 - k_1 - (k_2 - k_1)\frac{\eta}{2}\right)\tau_{2}$$

since  $I_{SLO,2}^{US,\mathcal{B}_1=\varnothing} = 0$  and  $I_{BMO,2}^{IH,\mathcal{B}_1=\varnothing} = 0$ . However, since  $\frac{k_3-k_1}{k_3} < \theta$ , then

$$\theta k_3 \tau + (1-\theta) \delta^2 \left( k_3 - k_1 - (k_2 - k_1) \frac{\eta}{2} \right) \tau > (k_3 - k_1) \tau$$

is always satisfied and, hence, in this case we conclude that in this case there is no equilibrium in which (BMO, SMO, BLO, SLO) is the strategy profile chosen at t = 1.

Fifth step. Based on the above, nobody at t = 1 has unilateral incentives to deviate whenever

$$\theta \leq \frac{k_3 - k_1}{k_3},$$

$$k_3 - k_1 \geq \delta \frac{\eta}{2} \left( k_3 + k_1 - 1 \right),$$

$$k_3 - k_1 \geq \theta k_3 + (1 - \theta) \delta^2 \left( k_3 - k_1 - (k_2 - k_1) \left( \lambda \pi + \frac{\eta}{2} \right) \right), \text{ and }$$

$$(\lambda \pi + \eta) \left( k_1 - 1 \right) - \lambda \pi k_3 > 0.$$

 $\mathcal{E}_2^D$ : (BMO, SMO, NT, NT)Following the same procedure we obtain that in this case  $Y = p \in [0, 1]$ ,  $X = \frac{\lambda \pi}{\eta + \lambda \pi}$  and the optimal responses of uninformed traders are:

State of the book	UB	US
$(A_1^1, B_1^1)$	NT	NT
$(A_1^2, B_1^1)$	$\begin{cases} NT & \text{if } Xk_3 \leq \frac{k_2 - k_1}{2} \\ BDO & \begin{cases} \text{if } \frac{k_2 - k_1}{2} < Xk_3 \leq k_2 \text{ or} \\ \text{if } k_2 < Xk_3 \text{ and } \theta > \theta_X \end{cases} \\ BMO & \text{if } k_2 < Xk_3 \text{ and } \theta \leq \theta_X \end{cases}$	$\begin{cases} SDO & \text{if } Xk_3 < \frac{k_2 - k_1}{2} \\ NT & \text{if } \frac{k_2 - k_1}{2} \le Xk_3 \end{cases}$
$(A_1^1, B_1^1 + \tau)$	$\begin{cases} NT & \text{if } Y \leq \frac{1}{2k_3} \\ BDO & \begin{cases} \text{if } \frac{1}{2k_3} < Y < \frac{k_1}{k_3} \text{ or} \\ \text{if } Y \geq \frac{k_1}{k_3} \text{ and } \theta > \theta_Y \end{cases} \\ BMO & \text{if } Y \geq \frac{k_1}{k_3} \text{ and } \theta \leq \theta_Y \end{cases}$	$\begin{cases} SDO & \text{if } Y < \frac{1}{2k_3} \\ NT & \text{if } \frac{1}{2k_3} \le Y \end{cases}$
$(A_1^1, B_1^2)$	$\begin{cases} BDO & \text{if } Xk_3 < \frac{k_2 - k_1}{2} \\ NT & \text{if } \frac{k_2 - k_1}{2} \le Xk_3 \end{cases}$	$\begin{cases} NT & \text{if } Xk_3 \leq \frac{k_2 - k_1}{2} \\ SDO & \begin{cases} \text{if } \frac{k_2 - k_1}{2} < Xk_3 \leq k_2 \text{ or} \\ \text{if } k_2 < Xk_3 \text{ and } \theta > \theta_X \\ SMO & \text{if } k_2 < Xk_3 \text{ and } \theta \leq \theta_X \end{cases}$
$(A_1^1 - \tau, B_1^1)$	$\begin{cases} BDO & \text{if } Y < \frac{1}{2k_3} \\ NT & \text{if } \frac{1}{2k_3} \le Y \end{cases}$	$\begin{cases} NT & \text{if } Y \leq \frac{1}{2k_3} \\ SDO & \begin{cases} \text{if } \frac{1}{2k_3} < Y < \frac{k_1}{k_3} \text{ or} \\ \text{if } Y \geq \frac{k_1}{k_3} \text{ and } \theta > \theta_Y \\ SMO & \text{if } Y \geq \frac{k_1}{k_3} \text{ and } \theta \leq \theta_Y \end{cases}$

Table B.6: Optimal responses of uninformed traders at t = 2 when the strategy profile at t = 1 is (BMO, SMO, NT, NT).

## $\mathcal{E}_{3}^{D}\text{: }(BLO,SLO,BLO,BLO)$

Following the same procedure we obtain that in this case  $Y = \pi$  and the optimal responses of uninformed traders are:

State of the book	UB	US
$(A_1^1, B_1^1)$	NT	NT
$(A_1^2, B_1^1)$	NT	SDO
$(A_1^1, B_1^1 + \tau)$	$\begin{cases} NT & \text{if } Y \leq \frac{1}{2k_3} \\ BDO & \begin{cases} \text{if } \frac{1}{2k_3} < Y < \frac{k_1}{k_3} \text{ or} \\ \text{if } Y \geq \frac{k_1}{k_3} \text{ and } \theta > \theta_Y \\ BMO & \text{if } Y \geq \frac{k_1}{k_3} \text{ and } \theta \leq \theta_Y \end{cases}$	$\begin{cases} SDO & \text{if } Y < \frac{1}{2k_3} \\ NT & \text{if } \frac{1}{2k_3} \le Y \end{cases}$
$(A_1^1, B_1^2)$	BDO	NT
$(A_1^1 - \tau, B_1^1)$	$\begin{cases} BDO & \text{if } Y < \frac{1}{2k_3} \\ NT & \text{if } \frac{1}{2k_3} \le Y \end{cases}$	$\begin{cases} NT & \text{if } Y \leq \frac{1}{2k_3} \\ SDO & \begin{cases} \text{if } \frac{1}{2k_3} < Y < \frac{k_1}{k_3} \text{ or} \\ \text{if } Y \geq \frac{k_1}{k_3} \text{ and } \theta > \theta_Y \\ SMO & \text{if } Y \geq \frac{k_1}{k_3} \text{ and } \theta \leq \theta_Y \end{cases}$

Table B.7: Optimal responses of uninformed traders at t = 2 when the strategy profile at t = 1 is (BLO, SLO, BLO, BLO).

 $\mathcal{E}_4^D$ : (BLO, SLO, NT, NT)

Following the same procedure we obtain that in this case Y = 1 and the optimal responses of uninformed traders are:

State of the book	UB	US
$(A_1^1, B_1^1)$	NT	NT
$(A_1^2, B_1^1)$	NT	SDO
$(A_1^1, B_1^1 + \tau)$	$\begin{cases} BDO \text{ if } \theta > \frac{k_3 - k_1}{k_3 - \frac{1}{2}} \\ BMO \text{ if } \theta \le \frac{k_3 - k_1}{k_3 - \frac{1}{2}} \end{cases}$	NT
$(A_1^1, B_1^2)$	BDO	NT
$(A_1^1 - \tau, B_1^1)$	NT	$\begin{cases} SDO \text{ if } \theta > \frac{k_3 - k_1}{k_3 - \frac{1}{2}} \\ SMO \text{ if } \theta \le \frac{k_3 - k_1}{k_3 - \frac{1}{2}} \end{cases}$

Table B.8: Optimal responses of uninformed traders at t = 2 when the strategy profile at t = 1 is (BLO, SLO, NT, NT).

## $\mathcal{E}_5^D$ : (BDO, SDO, BLO, SLO)

Following the same procedure we obtain that in this case the optimal responses of uninformed traders are:

State of the book	UB	US
$(A_1^1, B_1^1)$	NT	NT
$(A_1^2, B_1^1)$	NT	SDO
$(A_1^1, B_1^1 + \tau)$	NT	SDO
$(A_1^1, B_1^2)$	BDO	NT
$(A_1^1 - \tau, B_1^1)$	BDO	NT

Table B.9: Optimal responses of uninformed traders at t = 2 when the strategy profile at t = 1 is (BDO, SDO, BLO, SLO).

## $\mathcal{E}_6^D$ : (BDO, SDO, NT, NT)

Following the same procedure we obtain that in this case  $Y = p \in [0, 1]$  and the optimal responses of uninformed traders are:

State of	UB	US
$(A_1^1, B_1^1)$	NT	NT
$(A_1^2, B_1^1)$	NT	SDO
$(A_1^1, B_1^1 + \tau)$	$\begin{cases} NT & \text{if } Y \leq \frac{1}{2k_3} \\ BDO & \begin{cases} \text{if } \frac{1}{2k_3} < Y < \frac{k_1}{k_3} \text{ or} \\ \text{if } Y \geq \frac{k_1}{k_3} \text{ and } \theta > \theta_Y \\ BMO & \text{if } Y \geq \frac{k_1}{k_3} \text{ and } \theta \leq \theta_Y \end{cases}$	$\begin{cases} SDO & \text{if } Y < \frac{1}{2k_3} \\ NT & \text{if } Y \ge \frac{1}{2k_3} \end{cases}$
$(A_1^1, B_1^2)$	BDO	NT
$(A_1^1 - \tau, B_1^1)$	$\begin{cases} BDO & \text{if } Y < \frac{1}{2k_3} \\ NT & \text{if } Y \ge \frac{1}{2k_3} \end{cases}$	$\begin{cases} NT & \text{if } Y \leq \frac{1}{2k_3} \\ & \qquad \qquad$

Table B.10: Optimal responses of uninformed traders at t = 2 when the strategy profile at t = 1 is (BDO, SDO, NT, NT).

#### 

**Proposition 5** If  $k_1 = 1$ , then a PBE of the game is as follows.

• (BMO, SMO, NT, NT) is the optimal strategy profile for traders at t = 1 if

Conditions	
$\theta \le \frac{k_3 - k_1}{k_3}$	$k_3 - 1 \ge \frac{\eta}{2} \delta k_3,$
	$k_3 - 1 \ge \theta k_3 + (1 - \theta) \delta^2 \left( k_3 - 1 - (k_2 - 1) \left( \lambda \pi + \frac{\eta}{2} \right) \right).$
$\frac{k_3 - k_1}{k_3} < \theta$	Not an equilibrium.

The beliefs of uninformed traders at t = 2 are:  $X = \frac{\lambda \pi}{\eta + \lambda \pi}$ ,  $Y = p \in [0, 1]$  and  $Z = q \in [0, 1]$ . The optimal strategy for uninformed and informed traders at t = 2 are described in Tables B.6 and B.1 of Appendix B, respectively.

Conditions		
$ heta \leq rac{k_3-k_1}{k_3}$	$\frac{\eta}{2}\delta k_3 \ge \theta k_3 + (1-\theta)\delta^2 \left( (k_3 - 1) - (k_2 - 1) \left( \lambda \pi + \frac{\eta}{2} \right) \right),$	
	$\frac{\eta}{2}\delta k_3 > (k_3 - 1) .$	
$\frac{k_3 - k_1}{k_3} < \theta \le \frac{k_3 - k_1 + 1}{k_3 + \frac{1}{2}}$	$\frac{\eta}{2}\delta k_3 \ge \theta k_3 + (1-\theta)\delta^2\left((k_3-1) - (k_2-1)\left(\frac{\eta}{2}\right)\right).$	
$\frac{\frac{k_3 - k_1 + 1}{k_3 + \frac{1}{2}} < \theta$	$\delta \frac{\eta}{2} k_3 \ge \theta k_3 + (1-\theta) \delta^2 \left( (k_3 - 1) - \frac{\eta}{2} (k_2 - 1) \right).$	

• (BLO, SLO, NT, NT) is the optimal strategy profile for traders at t = 1 if

The beliefs of uninformed traders at t = 2 are: X = 0, Y = 1 and  $Z = q \in [0, 1]$ . The optimal strategy for uninformed and informed traders at t = 2 are described in Tables B.8 and B.1 of Appendix B, respectively.

• (BDO, SDO, NT, NT) is the optimal strategy profile for traders at t = 1 if

Conditions		
$ heta \leq rac{k_3-k_1}{k_3}$	$\theta k_3 + (1-\theta)\delta^2 \left( (k_3 - 1) - (k_2 - 1) \left( \lambda \pi + \frac{\eta}{2} \right) \right) > (k_3 - 1),$	
	$\theta k_3 + (1-\theta)\delta^2 \left( (k_3-1) - (k_2-1) \left( \lambda \pi + \frac{\eta}{2} \right) \right) > \frac{\eta}{2} \delta k_3.$	
$\frac{k_3 - k_1}{k_3} < \theta \le \frac{k_3 - k_1 + 1}{k_3 + \frac{1}{2}}$	$\theta k_3 + (1-\theta)\delta^2 \left( (k_3-1) - (k_2-1) \left( \frac{\eta}{2} \right) \right) > \frac{\eta}{2}\delta k_3.$	
$\frac{\frac{k_3 - k_1 + 1}{k_3 + \frac{1}{2}} < \theta}{$	$\theta k_3 + (1-\theta)\delta^2 \left( (k_3-1) - (k_2-1)\left(\frac{\eta}{2}\right) \right) > \frac{\eta}{2}\delta k_3.$	

The beliefs of uninformed traders at t = 2 are: X = 0,  $Y = p \in [0,1]$  and Z = 1. The optimal strategy for uninformed and informed traders at t = 2 are described in Tables B.10 and B.1 of Appendix B, respectively.

**Proof.** The Corollary follows immediately by taking the limit  $\pi$  goes to zero in the conditions that guarantee the existence of the PBE stated in Proposition 2.